

Delft University of Technology

A critical assessment and summary on the low carbon energy pile technologies based on the life-cycle perspective

Challenges and prospects

Zhang, Linfeng; Han, Haozhe; Li, Wenxin; Guo, Kewei; Yuan, Minglu; Liu, Zhengxuan

DOI 10.1016/j.applthermaleng.2024.122605

Publication date 2024 **Document Version** Final published version

Published in Applied Thermal Engineering

Citation (APA)

Zhang, L., Han, H., Li, W., Guo, K., Yuan, M., & Liu, Z. (2024). A critical assessment and summary on the low carbon energy pile technologies based on the life-cycle perspective: Challenges and prospects. Applied Thermal Engineering, 243, Article 122605. https://doi.org/10.1016/j.applthermaleng.2024.122605

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

Green Open Access added to TU Delft Institutional Repository

'You share, we take care!' - Taverne project

https://www.openaccess.nl/en/you-share-we-take-care

Otherwise as indicated in the copyright section: the publisher is the copyright holder of this work and the author uses the Dutch legislation to make this work public.



Contents lists available at ScienceDirect

Applied Thermal Engineering



journal homepage: www.elsevier.com/locate/apthermeng

Research Paper

A critical assessment and summary on the low carbon energy pile technologies based on the life-cycle perspective: Challenges and prospects

Linfeng Zhang^a, Haozhe Han^b, Wenxin Li^{c,*}, Kewei Guo^a, Minglu Yuan^a, Zhengxuan Liu^{d,e,*}

^a School of Transportation, Southeast University, 211189 Nanjing, China

^b Faculty of Construction and Environment, The Hong Kong Polytechnic University, 999077, Hong Kong, China

^c School of Energy and Environment, Southeast University, 211189 Nanjing, China

^d College of Civil Engineering, Hunan University, Changsha 410082, China

e Faculty of Architecture and the Built Environment, Delft University of Technology, Julianalaan 134, 2628 BL Delft, Netherlands

ARTICLE INFO

Keywords: Shallow geothermal energy Energy piles Full life-cycle Low carbonization Optimize design Building carbon emissions

ABSTRACT

Energy piles, a technology integrating the heat exchange component within building pile foundations for shallow geothermal energy utilization, have proven economically efficient. They outperform conventional ground source heat pumps by mitigating additional borehole costs and space requirements. This paper systematically examines low-carbon considerations and optimization measures throughout the planning, design, construction, and operation stages of energy piles, considering the entire lifecycle. Furthermore, this paper discusses potential challenges associated with decarbonizing energy piles, offering solutions based on case studies and environmental impact assessments. Through a comprehensive critical review and analysis of existing knowledge, this paper presents a systematic theory and methodology for optimal decarbonization of energy piles, serving as a valuable resource for building practitioners and researchers in this field. The findings not only contribute to a solid theoretical foundation but also provide technical support for the advancement and application of energy pile systems.

1. Introduction

In recent years, the global climate crisis has gained increasing prominence, primarily attributed to the extensive burning of fossil fuels and the resultant emission of greenhouse gases [1,2,155]. Reports indicate that achieving the Paris Agreement goal of limiting global warming to below 2 °C by 2030 requires a system-wide transformation, necessitating a 30 % reduction in emissions [3]. The Global Status Report for Buildings and Construction emphasizes the significant role of the building sector [153,154]], which accounted for 34 % of total final energy consumption and 37 % of global operational energy and process-related CO₂ emissions in 2021 [4]. In China, the building and construction sector alone contributed to 50.6 % of CO₂ emissions [5]. Therefore, it is imperative to reduce energy consumption and greenhouse gas emissions within the building sector to effectively work towards the carbon neutrality target [6,7].

In addressing this challenge, numerous scholars have directed their focus towards harnessing geothermal energy—a clean and renewable energy source that presents a viable alternative to fossil fuels,

particularly in building heating, ventilation, and air conditioning (HVAC) systems [8–10]. Within the realm of shallow geothermal energy utilization, energy piles have emerged as a focal point of interest among scholars. Energy pile constitutes a closed-loop ground source heat pump with three primary components: an earth connection subsystem, a heat pump subsystem, and a heat distribution subsystem [11]. The heat exchange pipes of energy piles are intricately embedded within the building's foundation piles, strategically positioned at a depth where soil temperature remains relatively stable. This symbiotic arrangement between foundation piles and heat exchange pipes forms a geothermal exchanger facilitating heat exchange with the surrounding soil [12,13]. Compared to traditional HVAC technology, energy piles exhibit both environmental and economic advantages throughout their life cycle, despite initial installation costs. Their diminished reliance on conventional energy sources and high energy efficiency contribute significantly to the reduction of CO_2 emissions [14–16]. Furthermore, energy piles offer broader applicability compared to other clean energy technologies, such as solar, wind, and tidal energy, given that shallow geothermal energy is geographically unrestricted. The concrete piles utilized in energy piles boast high thermal conductivity and heat storage

https://doi.org/10.1016/j.applthermaleng.2024.122605

Received 29 June 2023; Received in revised form 28 November 2023; Accepted 29 January 2024 Available online 1 February 2024 1359-4311/© 2024 Elsevier Ltd. All rights reserved.

^{*} Corresponding authors at: College of Civil Engineering, Hunan University (Zhengxuan Liu). School of Energy and Environment, Southeast University (Wenxin Li). *E-mail addresses:* wenxinli@seu.edu.cn (W. Li), zhengxuanliu@hnu.edu.cn (Z. Liu).

Nomenclature	SOO Single-objective optimization
	TPT Thermal performance test
Abbreviations	WSHP Water source heat pump
CFAContinuous flight augerCOPCoefficient of performanceGAGenetic algorithmGAHEGround-air heat exchangerGHEGround heat exchangerMOOMulti-objective optimizationRSMResponse surface methodologyHAVCHeating, ventilation, and air conditioningLCALife cycle assessmentMOOMulti-objective optimizationNSGA-IINon-dominated Sorting Genetic Algorithm IIPCMsPhase change materialsPHCPrestressed high-intensity concreteSiCSilicon carbide	SymbolsdDiameterkThermal conductivity f_e Electricity-related CO_2 emission factor E_{elec} Electricity consumption f_n Natural gas-related CO_2 emission factor E_{gas} Natural gas consumption Q Heat exchange rate c_p Specific heat capacity v Velocity ρ DensityTTemperature

capabilities, enhancing heat transfer efficiency between the pile and the surrounding soil in contrast to conventional borehole heat exchangers [17]. In addition, the integration of heat exchanger pipes within the pile foundation eliminates the necessity for additional boreholes, thereby reducing drilling costs and minimizing underground space consumption [18].

As a competitive geothermal heat exchanger technology, energy piles have been the subject of extensive research, covering the development of heat transfer models [19], sizing methods [20,21], and evaluations of technical, environmental, and economic feasibility [22]. These research endeavors have played a pivotal role in improving the energy efficiency of both the energy pile system and the buildings they serve. Laboratory, field, and numerical studies have documented these advancements [23]. Furthermore, investigations into the decarbonization potential of energy piles have been conducted. For instance, Han et al. [24] discovered that energy piles can achieve an average reduction in carbon emissions ranging from 25 to 36 % across different climate zones in eight cities. In a study conducted by Akrouch et al. [16] in Texas, USA, primarily under cooling weather conditions, and the implementation of an energy pile system resulted in a noteworthy reduction of electricity consumption by 5573MWh and carbon emissions by 3932 metric tons over 30 years. These studies underscore the significant advantages of energy piles in reducing carbon emissions through the utilization of geothermal energy and the enhancement of energy efficiency.

Based on above-mentioned literature review, several research gaps can be identified: 1) Existing research lacks a systematic summary of the carbon reduction effect of energy piles from the life-cycle perspective. 2) Existing studies on the decarbonization potential of energy piles primarily focus on reducing operational carbon emissions, neglecting the embodied carbon emissions associated with the construction stage and building materials. 3) The challenges and future research directions for maximizing the decarbonization potential of energy piles have not been adequately discussed in the existing studies.

This study aims to address these gaps by providing a comprehensive analysis of decarbonization considerations for energy piles at different life cycle stages. Relevant literature on their design, construction, operation, and comprehensive evaluation will be reviewed. Additionally, the study will outline potential challenges and barriers in using energy piles for carbon reduction and suggest future research directions to further minimize their carbon footprint. The following sections are organized as follows: Section 2 introduces the physical configuration of energy piles and discusses the advantages and disadvantages of each type. Section 3 and Section 4 summarize the decarbonization considerations for the design and construction stage, while Section 5 analyzes carbon reduction methods for the operation and monitoring stage. Section 6 provides an overview of the evaluation of the decarbonization potential for energy piles. Finally, Section 7 discusses the challenges and future research areas. A conclusion is presented in Section 8. This study will serve as a guideline for the future design, construction, operation, and evaluation of energy piles, offering technical support for achieving carbon neutrality in the building industry.

2. Common classifications and key impact parameters

2.1. Common classification

Energy piles and conventional ground source heat pumps are distinguished primarily by their underground energy structure. Energy piles can be categorized into different types based on the characteristics of the underground structure, considering the pile type and the configuration of pipes within the pile. Common energy pile types encompass cast-in-place reinforced concrete piles [18,25–27], steel piles [28–31], and precast reinforced concrete piles [32,33]. The configuration of pipes within energy piles varies, including single U-shaped, W-shaped, double U-shaped, triple U-shaped, helical [34,35]. U-shaped and W-shaped energy piles have pipes bent accordingly, while helical energy piles feature a spiral-shaped pipe. In terms of the connection forms of pipes, they can be divided into parallel and series types [36]. A visual representation of these structures can be seen in Fig. 1. The selection of the appropriate energy pile type depends on the specific geological conditions, load requirements, construction costs, and other factors.

The various forms of pipes directly affect the thermal performance, economic viability, and construction feasibility of energy piles [37-40]. Park et al. [33] proposed that the triple U-shaped energy pile had a better performance than the W-shaped energy pile. Mehrizi et al. [37] conducted a cooling performance comparison among three energy pile types (e.g., single U-shaped, W-shaped, and W-shaped-all round) and concluded that W-shaped-all round energy pile exhibited the best heat transfer efficiency. Gao et al. [38] evaluated the heat transfer efficiency of different types of cast-in-place concrete energy piles, determining that triple U-shaped energy piles were the most efficient, while W-shaped energy piles excelled at moderate media flow rates. Zarrella et al. [39] analyzed a model and found that, under the same pile geometry conditions and with identical pile and pipe materials, the spiral-shaped pipe demonstrated superior thermal performance compared to the triple Ushaped pipe. Khandouzi et al. [40] simulated the performance of single U-shaped, W-shaped, spiral-shaped, and 6U-shaped heat exchangers at different depths, revealing that the 6U-shaped configuration had the highest efficiency, followed by helical-shaped, W-shaped, and single U-



Fig. 1. Schematic diagram of different types of pipe configuration in energy pile [34,35].

shaped.

Yoon et al. [41] conducted a thermal performance test (TPT) comparing prestressed high-intensity concrete (PHC) energy piles with W-shaped and spiral-shaped pipes. They found that the heat flux of the spiral-shaped energy pile is higher. However, the construction cost of Wshaped energy piles is approximately three times lower due to the more expensive unit price and longer required length of spiral-shaped pipes. Meanwhile, Luo et al. [34] assessed the cost-benefits ratio of four types of pipe (e.g., double U-shaped, spiral-shaped, double W-shaped, and triple U-shaped) based on the heating and cooling needs of buildings and energy prices, and concluded that the triple U-shaped ground heat exchanger (GHE) has the highest economic performance, followed by double U-shaped, spiral-shaped, and double-W-shaped. However, in practical engineering, the absolute value of the heat output is often more crucial than cost considerations. Additionally, the construction feasibility of different pipe configurations also varies. For example, multi-Ushaped pipes require more complex installation due to multiple inlet and outlet ports, while the installation of spiral-shaped pipes typically requires more manpower [25,26].

The observed variance in heat transfer efficiency among different pipe types primarily stems from differences in the contact area between the pipe and the concrete [42–45]. As the contact area increases, the heat transfer efficiency increases. Considering the thermal performance and construction feasibility, the potential application of spiral-shaped pipes appears promising. Nevertheless, the selection of pipe configuration should also consider economic factors and construction constraints. Further research is essential to optimize the design and performance of energy piles with different pipe configurations.

2.2. Key influencing factors

Numerous factors influencing the thermal performance of energy piles fall into distinct categories: environmental, design, construction, and operating factors, as outlined in Table 1.

Environmental factors are essential considerations for any project. While these factors, such as site climatic conditions [46–49], thermal properties of the surrounding soil [50–52], groundwater condition (e.g., groundwater velocity, groundwater seepage, and groundwater level. [53–58]), cannot be optimized, they must be carefully taken into account to ensure a reasonable design. In contrast, design factors offer optimization opportunities during the design process. This includes adjusting the volumetric flow rate of the heat transfer fluid and considering geometric factors like pile length, diameter, pipe number, pipe length, and diameter.

Construction factors mainly include the construction method during the construction process and the selection of materials. The thermal conductivity (including thermal conductivity of heat transfer fluid, and thermal conductivity of pile material) and other thermophysical properties of materials (e.g., latent heat) have an impact on the operational efficiency of energy piles.

Operating factors involve adjustments made during operation to

Table 1

```
Summary of factors affecting the thermal performance of energy piles.
```

Environmental factors	Site climatic conditions [46–49] Thermal properties of the surrounding soil [50–52] Groundwater condition [53–58]		
Design factors	Geometric factors	Pile length [46,59,60] Pile diameter [46,61,62] Pipe number [63–65] Pipe length [45,66] Pipe diameter [46,67,68] Pile group arrangement [58,69,70]	
Construction factors	Construction met Material factors	hod [71,72] Thermal conductivity of heat transfer fluid [61,73–75] Thermal conductivity of pile material [62,76,77] Thermal conductivity of pipe material [78,79] Other thermophysical properties [80,81]	
Operating factors	Operation patterns	Intermittent operation [82–85] Zoning operation [86]	

achieve changes in system performance. Scholars focus on studying operation patterns, considering both temporal and spatial dimensions. Current research emphasizes the impact of intermittent operation temporally, while spatially, discussions involve grouping energy piles and zoning operation under various operating conditions.

3. Decarbonization considerations in design stage

3.1. Optimize design method

In the energy pile optimization process, the pivotal task involves defining objectives, variables, and optimization methods within specific constraints [87]. Design optimization can be categorized into single-objective optimization (SOO) and multi-objective optimization (MOO), depending on the number of optimization objectives [19]. The common optimization objectives can be summarized according to the "4E" criteria (i.e., energy criteria, exergy criteria, economy criteria, and environment criteria) [88]. This section will assess the optimization of energy pile design from a decarbonization perspective. Specifically, it will review optimization strategies with "energy criteria" and "environmental criteria" as primary design objectives, examining both SOO and MOO perspectives.

3.1.1. Single-objective optimization

Alberdi-Pagola et al. [70] proposed an optimization strategy based on the desirability function approach to maximize the pile spacing through a local optimization algorithm. This strategy leads to a reduction in the number of piles required for the pile group to meet the thermal load demand of the building. Ahmed et al. [79,89] identified nine design parameters and analyzed their significance using fractional factorial uniform design of experiments. Subsequently, Haridy et al. [90] refined the parameter significance analysis based on this, employing response surface methodology (RSM) to optimize coolingdominated energy piles. They determined the significance of each design parameter by analyzing the effect of eight design parameters on the thermal conductivity of the energy pile. Cecinato and Loveridge [61] used Taguchi analysis to analyze the key parameters of energy piles and optimize the design of energy piles by determining the relative importance of each parameter. Nazmabadi et al. [91] proposed a thermal recovery system that incorporates an open-loop ground-air heat exchanger (GAHE) pipe in the spiral-shaped energy pile. They combined genetic algorithms (GA) with response surface methods to optimize this system. In the optimization process, they used spiral pitch, pipe diameter, GAHE diameter and GAHE air inlet velocity as the main design parameters to optimize the maximum cooling load of per unit depth of energy pile provided by the overall system.

Numerous studies optimize design by simulating operational performance under different sets of design parameters. Bezvan et al. [92] compared the heat transfer efficiency of spiral energy piles with pitch of 0.2 m, 0.4 m, and 0.6 m, respectively. Carotenuto et al. [62] investigated the heat transfer performance of energy piles through numerical simulations using three geometry configurations of probes (i.e., double Ushaped, triple U-shaped probes and spiral coils) developed with COM-SOL Multiphysics and their performance was compared under different physical properties of the pile, geometrical characteristics of probes and fluid flow rate. After the traditional method of improving the heat transfer efficiency of energy piles by optimizing their geometric design configuration reached a bottleneck, some innovative optimizations were proposed. To improve the thermal conductivity of concrete, graphite was added to the concrete. Through laboratory testing and numerical simulations, Li et al. [93] compared the thermal efficiency of concrete energy piles containing different contents of graphite. Elkezza et al. [94] added different proportions of graphite powder into concrete and compared the thermal conductivity and the compressive strength of graphite-concrete with normal concrete. It was found both the thermal conductivity and the compressive strength of graphite-concrete were significantly improved. In addition to that, in Ref. [95-100], various novel pipe forms and their parameters are investigated to improve the thermal performance of energy piles. However, as no specific optimization method is employed, these optimizations often do not consider all possible solutions, and the outcomes may not be optimal.

3.1.2. Multi -objective optimization

As previously mentioned, MOO involves multiple optimization objectives. Based on the findings of Ref. [93], the possible effect on the strength of concrete with the addition of graphite was considered by Li et al. [101]. They made another attempt to add silicon carbide (SiC) to the concrete. They found that the SiC concrete not only increased in thermal conductivity but also in strength as the content of SiC increased. Meng et al. [102] developed an ontology-based decision support system for multi-optimal design of energy pile systems. The system prioritizes equipment cost, cost recovery period, CO₂ emission reduction and vertical loading capacity as the key design indicators. By using this system for multi-objective optimization design of cases, the authors finally obtained eight different optimized design solutions. Farajollahi et al. [67] analyzed the sensitivity of parameters in three-helix-shaped energy piles using response surface models, and then optimized the system using Non-dominated Sorting Genetic Algorithm II (NSGA-II).

While multi-objective optimization is not yet widely used in energy piles, its potential for integrating energy efficiency and environmental considerations through suitable optimization strategies is valuable for the decarbonization application of energy piles. More details of these SOO and MOO studies reviewed and the major findings obtained are summarized in Table 2.

3.2. Optimized influencing factors

The design stage holds substantial influence over the environmental impact of energy piles [104]. As the project progresses, the ability to impact environmental performance diminishes [105]. This is particularly true considering the challenges of altering the design of energy piles once the pile foundation begins to bear the load of the superstructure. Consequently, it is crucial to judiciously set key parameters during the design stage to enhance overall efficiency and minimize the carbon footprint of buildings. This section will outline the key parameters that affect the thermal performance of energy piles, offering the guidances for energy pile design.

3.2.1. Heat transfer fluid

Inside the pipe, the primary parameter affecting thermal performance is the heat transfer fluid. The heat transfer fluid circulates in the pipes to transfer heat between the ground and the heat pump system. Differences in the thermal conductivity and volumetric flow rate of the heat transfer fluid can have an impact on the thermal performance of the energy pile.

Common heat transfer fluids include pure water, water with antifreeze, or saline solutions [106]. Antifreeze options include ethylene glycol or propylene glycol. Ethylene glycol boasts advantages such as a lower price, high thermal conductivity, and low viscosity compared to propylene glycol. On the other hand, propylene glycol is non-hazardous [107]. Antifreeze is added to prevent fluid freezing during heat transfer. However, it's crucial to consider that increasing the antifreeze content raises fluid viscosity, potentially reducing heat transfer efficiency [108]. Therefore, selecting an appropriate antifreeze ratio is essential based on actual requirements. Currently, innovative approaches to heat transfer fluids have been proposed [73–75]. Researchers add nanofluids into heat transfer fluids to enhance their heat transfer properties.

3.2.2. Pipe diameter and the velocity of heat transfer fluid

The volumetric flow rate of the fluid is regulated by the pipe diameter and fluid velocity combined [109]. Typically, even though pipes may have varying shapes, an increase in pipe diameter and fluid velocity leads to an elevation in the volumetric flow rate, as depicted in Fig. 2. The heat flux of pile increases with volumetric flow rate until a certain critical value is reached. It can be explained by the relationship between Reynolds number and Nusselt number [59]. However, the growth of heat flux is nonlinear [28,46]. A blind increase in flow rate can diminish the time available for heat exchange, eventually resulting in inadequate heat exchange between the fluid and the surrounding soil [67,101,102]. In addition, Ding et al. [46] also proposed that the optimal design volumetric flow rate of the heat transfer fluid is different in summer and winter. Therefore, determining a reasonable volumetric flow rate is essential to avoid insufficient heat exchange or low heat transfer rates.

3.2.3. Pipe number

In general, an increase in the number of pipes has demonstrated a substantial improvement to the heat exchange performance of the energy pile [63,64]. However, a higher number of pipes leads to a more pronounced temperature development in the pile, resulting in increased thermal interaction between pipes [65]. The benefits of the additional pipe are gradually diminishing [110]. There is a significant nonlinearity between the increase in the number of pipes and the increase in exchanged power, as shown in Fig. 3 [61,111].

3.2.4. Pipe length

For energy piles configured with U-shaped pipes, altering the length of the pipes is typically accomplished by adjusting the number of pipes, and the impact aligns with the aforementioned considerations. However, for energy piles configured with spiral-shaped pipe, increasing the

Table 2

Summary of the basic information reported in the studies considering SOO and MOO.

Ref.		Optimization objective	Optimization variables	Optimization technique	Main findings
[70]	SOO	minimum pile number	Pile spacing	Local optimization algorithm	By optimization, the number of energy piles required for the case in the literature can be reduced by 32 %.
[79]	SOO	thermal conductance	Number of tubes, Pile diameter, Tube diameter, Tube thickness, Tube location, Pile conductivity, Tube conductivity, Soil conductivity. Water flow rate	Fractional factorial uniform design	Energy piles exhibit larger steady-state thermal conductivity for larger number of tubes, tube diameter, tube spacing and pile thermal conductivity.
[90]	SOO	thermal performance of the energy pile	Number of tubes, Pile diameter, Inner tube diameter, Tube thickness, Pile thermal conductivity, Soil thermal conductivity, Tube thermal conductivity, Water velocity	RSM	Inner diameter of U-tubes, pile thermal conductivity and tube thermal conductivity have significant influence on the thermal conductivity of the pile. The increase in thermal conductivity of tube and pile, as well as the increase in inner diameter, can significantly increase the thermal conductivity of energy piles. Tube thickness and water velocity do not have little effect on the thermal conductivity of the pile. When the pile diameter of the energy pile is large, the increase in the number of pipes helps to improve the thermal conductivity.
[92]	SOO	heat transfer rate	Pitch sizes	N/A	Spiral-shaped energy pile with 0.4 m pitch size had higher efficiency in heat transfer rate than the pile with 0.2 m or 0.6 m pitch size.
[61]	SOO	energy efficiency	Pile diameter, Pile length, Concrete cover, Concrete thermal conductivity, Pipe number, Pipe diameter, fluid flow velocity.	Taguchi method	Total pipe surface area is the most important factor for energy efficiency. The flow rate of the heat transfer fluid has a small effect on the overall exchanged energy of the energy pile.
[91]	SOO	provided cooling load per unit depth of the energy pile	Spiral pitch, Pipe diameter, ground-air heat exchanger (GAHE) pipe diameter, GAHE air inlet velocity	RSM GA	The use of a recovery system can increase the maximum provided cooling load of the system by 50 %. The 15-year average annual COP of energy piles with double GAHE pipe and single GAHE pipe recovery systems is increased by 30 % and 16 %
[62]	SOO	heat transfer rate	Concrete density, Concrete thermal conductivity, Pile diameter, Probe diameter, Fluid flow rate, Probe shape	N/A	The heat transfer performance increases with the increase of thermal conductivity. Spiral-shaped energy piles have better thermal performance than single-U-shaped, double-U shaped and triple-U shaped energy pile. Within a certain range, an increase in pile diameter or pipe diameter can lead to an increase in heat transfer rate.
[103]	SOO	exchanged power of energy micro-piles	Fluid velocity, Pipe thermal conductivity, Fluid thermal conductivity, Pile diameter, Pile length, Concrete thermal conductivity, Pipe diameter	Taguchi method	The design of energy micro-piles is different from the design of energy piles. the pipe diameter is a key factor in the design of energy micro-piles. Among the site factors, the extent of thermal insulation between the ground and the building is also very important, in addition to the ground thermal conductivity.
[93]	SOO	total thermal resistance	Graphite volumetric contents	N/A	Graphite can improve the heat transfer coefficient, with a graphite content preferably higher than 15 %.
[101]	МОО	thermal conductivity of concrete the strength of the concrete	SiC contents Graphite contents	N/A	Although the increase of graphite content can make concrete thermal conductivity increase, however, the strength of concrete will be significantly reduced. The average thermal conductivity reaches 2.92 W/(m·k) and the compressive strength of concrete is 28.7 MPa when the graphite content reaches 15 %, indicating a reduction of 29.8 % in strength (The compressive strength of block without graphite is 42.3 MPa). When the SiC content is 15 %, the thermal conductivity reached 2.47 W/ (m·k) and the compressive strength was increased by 6.8 %.
[67]	MOO	the ratio of the load provided to the load required by the building the system cost	Helix pitch, Helix diameter, Pipe diameter	RSM NSGA II	The helix pitch is the most key parameters influencing the thermal performance, followed by helix diameter and the pipe diameter.

length of pipe within a fixed pile length can also be achieved by narrowing the pitch. It's important to note that while narrowing the pitch enhances the exchanged power of the pile, it concurrently diminishes the heat flux of the pipe due to the adverse interactions between pipes [45,62,66], as illustrated in Fig. 4.

3.2.5. Pile length

Pile length exerts a substantial influence on the thermal performance of energy piles [59–61]. The exchanged power of the pile markedly

increases with a longer pile length due to the expanded contact area. However, it's crucial to note that the thermal efficiency per unit area cannot be enhanced indefinitely, as depicted Fig. 5. Furthermore, the selection of pile length is more contingent on the load-bearing requirements of the pile than on the thermal properties.

3.2.6. Pile diameter

In general, larger pile diameters improves energy efficiency of energy piles [61]. This is because a larger diameter allows for a reasonable



Fig. 2. The heat exchange rate and volumetric flow rate of the fluid: (a) for different fluid velocity [109]; (b) for different pipe diameters [109].

spacing of the internal pipes, and the mutual thermal interference between the pipes affects the overall efficiency of the energy pile [62], see Fig. 6 (a). In addition, a larger diameter also provides the possibility of placing more pipes in the same pile, resulting in the improvement of the efficiency of a single pile. Ding et al. [46] found that except for individual months, the maximum exchanged power increases with the increase of pile diameter, but the difference is not significant, as shown in Fig. 6 (b).

3.2.7. Comprehensive analysis

Currently, there is limited literature on the comprehensive analysis of various parameters of energy piles. Cecinato and Loveridge [61] conducted a thorough analysis of pile length, pile diameter, pipe diameter, number of pipes, heat transfer fluid flow rate, concrete thermal conductivity, and thickness of concrete cover and ranked the importance of these parameters, see Fig. 7 (a). Ahmed et al. [79] defined nine factors: number of pipes (n), pile diameter (d_p), pipe diameter (d_i), pipe thickness (T), pipe spacing (S), pile conductivity (K_p), pipe conductivity (K_t), soil conductivity (K_s), and heat transfer coefficient of the circulating water (H). The effects of these parameter can be observed in Fig. 7 (b).

From the comprehensive comparison of various parameters, it is evident that the number of pipes and pile length have the maximum impact on the thermal performance of energy piles. However, the



Fig. 3. The effect of number of pipes on the thermal performance of energy pile: (a) The exchanged power for different number of pipes [61]; (b) Energy extracted by single u-pipe and double u-pipe within 28 days [111].

determination of pile length is contingent on geological conditions and bearing requirements in practical applications. Therefore, optimizing the number of pipes is crucial for improving the thermal performance of energy piles. It should be noted that although the number of pipes can be independently determined, blindly increasing the number of pipes is not feasible. The limitations of pile diameter and the thermal interference between pipes need to be considered in the design. In addition, from a cost-benefit perspective, the benefits of additional pipes decrease with the addition of more pipes. The effect of the thermal resistance of the concrete cannot be ignored either. However, optimizing this parameter is challenging as the thermal resistance of concrete depends on its aggregate characteristics and mix ratio, and the primary function of the pile body is for bearing services. Based on the earlier analysis of the impact of pile diameter, although its effect on the thermal performance of energy piles is limited, prioritizing larger pile diameters in the design process is advisable. While reducing the thickness of the concrete



Fig. 4. The effect of pipe length on the thermal performance of energy pile: (a) Exchanged power (left axis) and heat transfer rate per unit of length of pipe (right axis) with different pipe length [62]; (b) Heat of flux of pile along with different heat exchange pipe length [66].

protective layer is advantageous for the heat exchange of pipes, the specific thickness setting should consider the thermal–mechanical effects of the pile body comprehensively.

Furthermore, when considering the influence of the key parameters mentioned above on the heat transfer performance of energy piles, it becomes evident that an augmentation in the contact area between the pipe and the concrete pile positively contributes to the improvement of heat transfer performance [42–45]. The increase in the number [40,61], diameter [67], and length of pipe [66] can all improve heat transfer performance by increasing the contact area.

3.2.8. Pile group arrangement

Additionally, beyond the design of individual piles, the rational design of pile groups is a crucial aspect of energy pile systems. According to You et al. [69], the efficiency of the energy pile group with a spacing of 7 m is higher than that with a spacing of 3 m. Regarding pile layout, a linear arrangement of energy piles proves more efficient than a stripe-shaped pile group or a rectangular pile array. Alberdi-Pagola et al. [70] redesigned an existing pile group by simultaneously considering the maximization of the distance between piles and the minimum number of energy piles required for the building's thermal load, reducing the number of piles in the redesigned pile group by about 1/4,



Fig. 5. The effect of length of pile on the thermal performance of energy pile: (a) Total exchanged power vs pile length at different interface [59]; (b) Exchanged power normalized by pile surface area vs pile length at different interface [59].

and significantly improving the efficiency of individual piles. Mehrizi et al. [37] proposed that different pipe-shaped energy piles have different influence radius, and during construction, the radius beyond the determined range should be exceeded to avoid mutual interference. In their study, the influence radius of a single-U shaped energy pile was 1.5 m, and that of a W-shaped energy pile was 2.5 m. Lyu et al. [68] found that the heat transfer rate of corner piles in clover and square pile

group layouts was 6.8 % and 9.9 % higher than that of central piles. If only considering the thermal performance, they suggest that the pile spacing should be greater than 6.8 times the pile diameter. Tiwari et al. [112] proposed that when normalized spacing $s/d \le 15$ (s is center-to-center spacing between two piles, d is the diameter of the pile), the power output from the pile group is less than the cumulative power output from the same number of isolated piles.



Fig. 6. The effect of length of pile on the thermal performance of energy pile: (a) Total exchanged power and heat transfer rate per unit length of pipe for different pile diameter [62]; (b) The monthly heat exchange for the different pile diameter [46].

Pile group layouts have a significant impact on the overall system and the efficiency of individual pile. Considering the mutual influence between energy piles, the pile spacing and pile layout should be given priority in the design. The spacing between thermal-active piles should exceed the influence radius wherever possible. On the other hand, the balance between the thermal efficiency of the single pile and the total amount of energy supplied also needs to be considered. To meet the total thermal demand of the building, the thermal efficiency of the single pile may be sacrificed for the placement of more thermal-active piles, increasing the total exchanged power.

4. Decarbonization considerations in construction and demolition stage

4.1. Construction method

Throughout the lifecycle of energy piles, the construction phase significantly impacts overall efficiency and economic costs. A reduction in efficiency necessitates more conventional energy during operation, diminishing the carbon reduction potential of energy piles.

Currently, scholars have summarized the installation methods of geothermal loops for various pile foundation technologies. For rotary



Fig. 7. The importance of different parameters on the thermal performance of energy pile: (a) The ranking of importance of parameter [61]; (b) The effect of the investigated parameters on energy pile conductance [79].

drilling piles, Amis et al. [71] indicated that the heat exchanger pipes can be installed inside the reinforcing steel cage, and the pipes can be allowed to hang below the cage to fully utilize the depth of the pile. During construction, a small amount of additional steel reinforcement can prevent the problem of pipes floating and not being fully utilized due to their buoyancy. For driven cast in-situ piles, Amis et al. [71] proposed that it is necessary to add a steel reinforcement throughout the length of the pile to avoid the problem of concrete "necking down" during casing withdrawal. Loveridge and Cecinato [72] discussed the construction differences and measures for continuous flight auger (CFA) piles, where steel cages are placed after pumping the borehole with a helical auger. This limits pipe length to the steel cage's range, requiring separate installation for full-depth requirements. Park et al. [26] evaluated the feasibility of installing multiple pairs of pipelines in large-diameter castin-place concrete piles. Large-diameter piles provide more pipeline space, but due to the limited surface area of the reinforcing cage, they can be installed on the outer and inner circumferences of the reinforcing frame separately. When installing multiple pairs of pipes, the complicated pipe connection at the pile head needs to be considered. During construction, it is necessary to distinguish the inlet and outlet of each pair of U-shaped heat exchange pipe. Fig. 8 and Fig. 9 summarize the common construction processes for energy piles. Ensuring quality in each construction phase is crucial for efficient energy pile operation, subsequently reducing carbon emissions during operation.

The demolition stage of energy pile has not been extensively studied. However, recycling and reusing waste from energy pile demolition could



Fig. 8. Construction flow chart of energy pile system.

be a viable measure for further reducing carbon emissions in line with decarbonization goals.

4.2. Construction materials

During the construction phase, the selection of building materials for energy piles to reduce carbon emissions primarily involves enhancing



(a) Setting out



(b) Installing temporary casing



(c) Boring



(d) Air-lifting



(g) Grouting



(e) Attachment of heat exchange pipe



(h) Removing casing



(f) Placing of steel cage



(i) After construction

Applied Thermal Engineering 243 (2024) 122605

the operational efficiency of energy piles and minimizing the embodied carbon of the materials.

Current research focuses on utilizing phase change materials (PCMs) to enhance the operational efficiency of energy piles. Bao et al. [80,113] encapsulated phase change paraffin in hollow steel balls within concrete, creating energy piles with PCM. Their findings indicate that PCM energy piles consistently exchange more power than traditional concrete energy piles. Based on this, Cui et al. [114] explored further, noting that PCM could reduce the water absorption capacity of concrete. To address this, they added steel fibers to PCM concrete, limiting water absorption while improving thermal conductivity, resulting in a 71 % increase in thermal conductivity. Cao et al.[81,83] employed the Taguchi-Grey relational analysis method to investigate the impact of PCM in precast high-strength concrete energy piles. They concluded that the increased latent heat and thermal conductivity of PCM could enhance the thermal efficiency of PCM precast high-strength concrete energy piles. However, Mousa et al.[115] pointed out that when PCM reaches a complete state, it may have a negative impact on the thermal performance of energy piles. The location of PCM in optimizing energy piles is also a crucial consideration.

In addition to using PCM to improve the thermal efficiency of energy pile operation and reduce operational carbon emissions, scholars also aim to minimize the embodied carbon in concrete piles. Shen et al. [77] introduced alkali-activated concrete in energy piles, observing improvements in thermal properties and a 32 % reduction in CO₂ emissions compared to Portland cement concrete energy piles. Alkali-activated concrete energy piles also extracted 17 % more thermal energy. While extensive studies on low-carbon energy piles have been conducted,

Fig. 9. Construction flow chart of energy pile system [25].

research on low-carbon concrete remains in its early stages. The thermomechanical properties of low-carbon concrete with different additives still require exploration [116–121] (see Table 3). Investigating the thermomechanical properties of low-carbon concrete and assessing the economic feasibility of using it in energy piles are crucial steps toward further reducing the embodied carbon of piles.

5. Decarbonization consideration in operation stage

5.1. Monitor method

To address the scarcity of quantitative analysis regarding the carbon reduction impact of energy piles, this section aims to consolidate methods for evaluating such reductions based on existing literature, and it will outline the essential monitoring data required for precise quantitative analysis.

The total CO₂ emissions caused by the operation of HVAC systems can be calculated based on the local energy structure. Some studies [15,24] consider CO₂ emissions related to electricity consumption and natural gas combustion, with a specific study [16] examining the reduction in CO₂ emissions attributed to variations in electricity consumption. The CO₂ emissions of HVAC systems can be calculated by the Eq. (1) [24].

$$Total \ CO_2 \ Emission = f_e \times E_{elec} + f_n \times E_{gas}$$
(1)

where f_e is the electricity-related CO₂ emission factor and E_{elec} is the electricity consumption, f_n is the natural gas-related CO₂ emission factor and E_{gas} is the natural gas consumption.

In an energy pile system, the energy flux received by the secondary circuit comprises a blend of energy extracted from the ground and the external power required to operate the heat pump (Fig. 10). This configuration enables the energy output after the heating pump to surpass the energy input for operation, implying that more energy can be supplied while consuming the same amount of electricity. Generally, the thermal performance of an energy pile system can be assessed using the coefficient of performance (COP).

Several studies on the COP of energy pile system have been collated,

 Table 3

 Summary of the basic information in the studies about the low-carbon concrete.

Ref.	Basic information	Conclusions
[116]	Supplementary cementitious materials (Fly ash)	Replacing 20 % clinker with glass cullet can reduce energy consumption by about 16 % and greenhouse gas emissions by 17 %.
[117]	Supplementary cementitious materials (Fly ash)	Fly ash can reduce the carbon dioxide emissions of concrete by 13 $\%$ to 15 $\%.$
[118]	Supplementary cementitious materials (Fly ash)	The CO_2 emission of concrete with 50 % ordinary Portland cement replaced by fly ash is 41.42 % lower than that of full cement concrete.
[119]	Low-carbon cementitious binders (Geopolymers)	Compared with ordinary Portland cement, geopolymer concrete has an estimated reduction of 44 % to 64 % in greenhouse gas emissions.
[120]	Low-carbon cementitious binders (Metakaolin geopolymer)	Compared to cement concrete, geopolymer concrete reduces global warming potential by 61 %.
[121]	Low-carbon cementitious binders (Calcium sulfoaluminate)	The direct CO_2 emissions of Calcium Sulfoaluminate clinker are 34 % lower than those of ordinary Portland cement clinker.

as depicted in Table 4. In Ref. [25,32,122–124], the COP of energy pile system typically ranges between 3.5 and 4.5, which means the secondary unit can obtain approximately four times more energy, with one portion of electrical input. The decrease in the reliance on fossil fuels for electricity generation underscores the carbon reduction benefits of energy piles.

In the process of calculating the emission reduction benefits of energy piles, the amount of energy heat exchange rate Q_{prim} provided by the primary circuit can be calculated by monitoring temperature sensors placed at the inlet and outlet, using Eq. (2)[113]. The power consumed by pile system P_{HP} can be measured by cumulative power meter. Then, the heat flux in secondary circuit can be calculated, using Eq. (3)[106]. The equivalent electricity demand of a traditional system can be evaluated based on Q_{sec} . After calculating the energy consumption of both the energy pile system and the traditional system, the CO₂ emission reduction can be calculated according to the above Eq. (1).

$$Q_{prim} = c_p \rho v \delta T = c_p \rho v (T_{out} - T_{in})$$
⁽²⁾

$$Q_{sec} = Q_{prim} + P_{HP} \tag{3}$$

where Q_{prim} is the heat exchange rate (W), c_p is the specific heat capacity of the circulating water (J/(kg °C)), ν is the velocity of the circulating water (m/s), ρ is the density of the circulating fluid (kg/m³), δT is the difference between the outlet temperature (T_{out}) and inlet temperature (T_{in}) (°C), Q_{sec} is the heat flux in secondary circuit (W), P_{HP} power consumed by pile system (W).

Other data crucial for assessing thermal performance includes monitoring surrounding soil temperature and heat transfer fluid flow over time. During the long-term operation of the energy pile system for heat extraction/injection, the loss/accumulation of underground heat can influence the underground temperature. Significant alterations in the initially stable ground temperature may result in decreased efficiency after several years of operation [126]. Additionally, it is essential to monitor and control the flow rate within a reasonable range. Flow rates that are excessively fast or slow can lead to insufficient heat exchange or low heat transfer rates.

5.2. Optimization operation

Currently, there is less focus on the operational optimization of energy systems compared to design optimization. Allen Bowers et al. [129] grouped the piles based on their location. By operating different groups of piles at different extraction rates and by running piles in groups from the outside to the inside, the ground temperature variation is optimized, avoiding thermal efficiency loss. You et al. [86] conducted numerical simulations to compare the thermal performance of the energy pile system under four zoning operation strategies. The study found that zoning operation effectively alleviates soil thermal imbalance, improving the system's COP. In Ref. [82-85,130-132], the improvement of the thermal performance of energy piles by different operation modes is explored. Compared to the conventional continuous operation mode, the intermittent operation mode (e.g.,16 h operation with 8 h rest, and 8 h operation with 16 h rest) allows for a higher heat exchange rate. At the same time, the intermittent mode of operation allows for less variation in ground temperature. The major findings of these operation optimization studies reviewed are presented in Table 5.

6. Decarbonization assessment

6.1. Case studies

The application of energy piles has significant implications for reducing CO_2 emissions. In current cases, energy piles are mostly used in commercial or office buildings. Simulating the long-term operation of the energy piles allows the calculation of the energy supplied by the system, and when combined with the local energy structure, it provides



Fig. 10. Basic principle of Thermal performance monitoring [106,127,128].

an estimate of the long-term carbon reduction potential of energy piles. This section reviews the quantification studies of energy pile emissions reduction to demonstrate the low-carbon characteristics of energy piles more intuitively.

The multi-functional hall (Europe) [106,108]: The multi-functional hall was designed with a capacity of 8000 persons, using 320 energy piles (18 m) for space heating and cooling. The total length of the heat exchange pipes is approximately 65 km. The system can save 85,000 m³ of natural gas per year, equivalent to a reduction of 73 tons of CO_2 emissions.

The low-energy shopping center (Austria) [106]: The shopping center used 650 thermal-activated piles (diameter D = 0.9 m, some piles have a diameter of D = 1.2 m, depth = 50 m). By using energy piles instead of traditional heating/cooling systems, the center saved nearly 550 tons of CO₂ emissions. Excess geothermal energy is transported to the public district energy supply line.

The Liberal Arts Building on the Texas A&M University Campus (Texas, USA) [16]: The five-story building covers a total area of 11,575 m² on a 2,885 m² land area, based on 263 foundation piles. The climatic condition is warm humid temperate climate. The temperature range is usually between 4.4 °C and 35 °C throughout the year. The warm season have a daily average high temperature of 31.4 °C or above, lasting from the end of May to the end of September. By monitoring the operation of the conventional HVAC system for one year, the total electricity consumption was determined to be 817,951 kWh. Simulations were performed for the operation of 263 energy piles (single-u pipe, pile spacing = 1.46 m, pile diameter = 0.45 m, pipe spacing = 0.2 m, pile length = 17.4 m, total pipe length = 4,734 m, $\lambda_{concrete}$ = 2.30 W/m.K λ_{soil} = 1.87 W/m.K) in the building for 30 years. The use of this system will reduce total electricity consumption by 5,573 MWh over 30 years. This reduction in

electricity consumption will lead to a total reduction of 3,932 tons of CO_2 emissions over 30 years.

A seven-story building (Naples, Milan, Italy) [15]: The building has a total heat transfer area of 1609 m^2 and a gross heated volume of 5991 m³. Operation in two different climate zones is considered for the same building: Naples in southern Italy with an external design temperature of 2.0 °C in winter, and Milan in northern Italy with an external design temperature of 5.0 °C. The temperature inside the building is assumed to be uniform, with a temperature of 20 °C during the heating season and 26 °C during the cooling season. The building has 24 energy piles, each 25 m long and consisting of four u-shaped pipes per pile, with a total pipe length of approximately 4800 m. A horizontal connection of about 500 m was also considered. The net energy required for heating and cooling are provided. In Naples, the cooling energy demand is about 39,500 kWh/year, while the heating energy demand is about 15,500 kWh/year. In Milan, the cooling energy demand is about 25,100 kWh/ year, while the heating energy demand is about 52,700 kWh/year. Based on a 20-year simulation, it was concluded that the use of energy pile systems can reduce greenhouse gas emissions by approximately 20 % for the building in Naples, and by no more than 10 % for the building in Milan.

In addition to the above cases, Han et al. [24] conducted a comparative study of energy pile system against a water source heat pump (WSHP) system + an evaporative fluid cooler + a natural gas boiler and an HVAC system that directly utilized regional heating and cooling as energy sources. Regardless of location and climatic zone, the energy pile system was found to emit the least amount of CO₂, as illustrated in Fig. 11. This research highlights the versatility and superiority of energy pile systems across different environments, resulting in a significant reduction in CO_2 emissions. Across the eight cities studied,

Table 4

Summary of the basic information in the studies about the COP of energy pile system.

Ref.	Basic information	COP
[25]	Pipe: Diameter: inner diameter = 21 mm outer diameter = 27 mm 5-pair-parallel U-type, Length = 130 m:8-pair-parallel U-type Length = 208 m 10-pair-parallel U-type Length = 260 m; S-type Length = 160 m Coil-type 500 mm coil pitch Length = 101 m Coil-type 200 mm coil pitch Length = 240 m	COP = 2.91; The system performance factor (SPF) = 2.78 (heating mode) COP = 3.40; SPF = 3.14 (cooling mode)
[32]	Pile: Length = 9 m; Diameter: inner diameter = 232 mm; outer diameter = 302 mm Pipe: Shape: U-shaped; Diameter inner diameter = 28.8 mm; outer diameter = 34 mm	COP = 3.9 SPF = 3.2
[122]	Pile: Length = 15 m; Diameter: inner diameter = 0.34 m; outer diameter = 0.5 m Pipe: Shape: Coil-shaped; Daimeter:25 mm Length = 196 m	COP = 3.9-4.3
[123]	Pile: Length = 10 m; Diameter = $0.3 m$ Pipe: Shape: U-shaped; Diameter: inner diameter = $0.032 m$; outer diameter = $0.013 m$	first year operation: the average COP = 3.40 (heating mode) the average COP = 4.63 (cooling mode) the tenth-year operation: the average COP = 3.82 (heating mode) the average COP = 4.31 (cooling mode)
[124]	150* reinforced concrete piles, 13 m long 10 * 180 m long pipes buried in the base slab	The COP of the heat pump units of the energy pile varies in the range of $4.2-4.7$ and the average daily COP of the system varies in the range of $3.5-3.9$.
[125]	Pile: Length = 10 m; Diameter = 0.8 m Pipe: Shape: spiral-shaped; Diameter inner diameter = 26 mm; outer diameter = 32 mm; Pitch = 0.2 m; Length = 85.3 m	COP = 4.7-4.8 (heating mode) COP = 4.4-4.7 (cooling mode)

energy piles achieved an average reduction in CO_2 emissions ranging from 25 % to 36 %.

The existing literature underscores the noteworthy advantage of the energy pile system in reducing greenhouse gas emissions, a critical aspect in its feasibility assessment. However, there is currently a scarcity of long-term monitoring data on the application of energy pile systems in buildings, necessitating further research on the emission reduction effects of energy pile systems in buildings situated in diverse climatic regions. Moreover, the present research on the emission reduction effects of energy piles predominantly centers on lowering operating carbon, with insufficient attention to embodied carbon. A comprehensive examination of both operating and embodied carbon can provide insights into the low-carbon performance of energy piles and guide further research directions for more effectively reducing carbon emissions. Energy piles, given their advantage of not requiring additional drilling compared to traditional ground source heat pumps, merit prioritized consideration in engineering projects. Governments should augment investments in energy piles, considering environmental factors such as greenhouse gas emissions, through policy and funding support.

6.2. Life cycle assessment

In the application of energy pile technology, it is crucial to not only assess energy efficiency during the operational stage but also consider its environmental impact throughout its entire lifecycle for a comprehensive evaluation of its engineering applicability.

Table 5

Summary of the basic information reported in the main optimization operation studies reviewed.

Ref.	Optimization objective	Optimization variables	Main findings
[129]	Minimize the amount of energy lost in the ground	Combinations of heat injection and extraction	Operating the single pile in the pile group from outside to inside depending on their location and operating the single pile in the pile group at different heat extraction rates (high extraction rate for the outer piles and low extraction rate for the inner piles) can result in less variation in ground
[86]	Alleviate the soil thermal imbalance, improve thermal performance of the system	Operational pile group	Continuous heat Continuous heat extraction will lead to the cold accumulation, which can be relieved by injecting dense heat into center of pile group or extracting heat from outer layer of the pile group. The heating COP can be improved from 3.297 to 3.432 by using zoning operation strateev
[82]	The thermal and mechanical responses of energy pile	Operation mode: 16 h-on, 8 h-off	The heat exchange rate under intermittent operation is 53.5 % higher than under continuous operation.In the intermittent operation, the accumulated temperature decrease is smaller. Intermittent operation induced lower thermal loads on the foundation and the mechanical response of the pile is reversible.
[83]	The heat exchange rate of energy pile backfilled with PCMs	Operation mode: 6 h-on, 18 h-off 8 h-on, 16 h-off 16 h-on, 8 h-off 18 h-on, 6 h-off	The heat exchange rate of energy pile tends to decrease as the operating time increases. The enhancement effect of intermittent operation on the heat transfer performance of the energy pile backfilled with PCMs is more obvious because PCMs can recover to their initial state under the intermittent pattern.
[84]	The system performance for the long-term operation	Operation mode: 10 h-on,14 h-off	The system maximum and minimum thermal energy output is smaller under the intermittent operating condition. Intermittent mode operation saves approximately 50 % of power per month compared to continuous mode operation. (From November to April, 43.4 %, 55.9 %, 55.6 %, 55.3 %, 46.9 % and 44.4 % respectively)

Table 5 (continued)

Ref.	Optimization objective	Optimization variables	Main findings
[85]	The thermal performance of the energy pile subjected to cooling	Operation mode: 8 h-on, 16 h-off 16 h-on, 8 h-off	From November to April, The mean monthly COPs of the intermittent operation mode are 3.63, 3.58, 3.45, 3.21, 3.25 and 3.34. The mean monthly COPs of the continuous operation mode are 3.32,3.27, 3.22, 3.03, 3.10 and 3.24. The intermittent mode of operation provides a significant improvement in average energy extraction. (8 h-on, 16 h- off: 40.9 % higher than continuous mode, 16 h- on, 8 h-off: 14.8 % higher than continuous mode) The shorter the running time, the lower the thermal impact on the ground, but the higher the average pile temperature during operation.



Fig. 11. Cost-saving, energy-saving, and CO₂ emission reduction of energy pile system [24].

For energy pile systems, the economic costs occur throughout their lifecycle, with energy savings realized during the operational stage. Environmental impacts can occur during production, installation, operation and demolition stage, as shown in Fig. 12 [133]. Sutman et al. [134] employed the life cycle assessment (LCA) methodology to evaluate the environmental performance of energy piles and compared it with that of conventional heating and cooling systems. The results demonstrated significant advantages of energy piles over conventional systems in four dimensions: equivalent CO_2 emissions, human health, resource consumption, and ecosystem quality. Furthermore, other LCA studies [133,135,136] focusing on geothermal energy for building heating and cooling have consistently shown that the application of geothermal energy is notably superior to conventional systems in terms of energy savings and environmental impact. It's important to note that

the primary energy structure for power generation in different regions can influence the results of environmental assessments.

7. Challenges and future research directions

7.1. Potential challenges to wider adoption of energy piles for carbon reduction

7.1.1. Design perspective

In the design process of energy piles, the temperature change of the pile is an important parameter. Due to the transfer of heat through the pile body of energy piles, there is spatial difference and non-uniformity in the temperature distribution of the pile cross-section [137]. However, current design methods often rely on the maximum temperature change to calculate the thermo-mechanical response of the pile. This approach not only overlooks the nuanced influence of temperature changes on the pile, providing an unrealistic description of energy piles [138], but may also cause design redundancy and increased design costs. In addition, many commercial design programs use simplistic methods to consider temperature responses [18]. Common analysis methods, such as the line source model and cylindrical source model, neglect factors like axial heat flow along depth, soil surface heat flow, and bottom heat flow [139]. The application of simplified analysis methods can constrain the long-term analysis of the system's operation.

In the process of load calculation, there is a difference between the assessment of heating and cooling demand for buildings and the actual requirements. Current methods mostly lead to underestimation of demand [140]. On the other hand, since the number and size of energy piles depend on the mechanical requirements of buildings, the supply of heat load may not fully meet the demand, necessitating consideration of hybrid systems. Existing research primarily focuses on the analysis of single piles or pile groups, lacking studies on the design of energy pile hybrid systems. In hybrid systems, challenges such as accurately determining the heat load that each system can provide and determining the appropriate control mode need to be addressed.

An integrated platform for energy pile design remains a critical need, requiring ongoing development and enhancement. Existing design methods for traditional energy pile systems are often guided by singular objectives, lacking the capability to holistically consider relevant factors across different domains. This limitation is tied to the simplification methods employed during design and certain specific numerical calculation software [102]. Energy pile design necessitates the simultaneous consideration of load-bearing requirements and heating/cooling demands. The heat transfer occurring during energy pile operation may induce changes in local mechanical properties. During long-term operation, both the mechanical and thermodynamic properties of energy piles will inevitably undergo certain changes [141,142]. In addition, research on optimization design of energy piles is currently insufficient. Integrating various algorithms for optimizing energy piles has the potential to enhance overall system performance.

7.1.2. Construction perspective

The construction of energy piles mainly involves pipeline placement and fixing operations. Compared with traditional borehole GHE systems, energy piles have larger diameters, providing geometric flexibility for the internal heat exchange tubes. However, the increased geometric complexity also increases the uncertainty of construction quality [143]. Improper operations may cause pipe rupture and leakage of heat exchange fluid. In addition, non-standard construction may lead to the bending or blockage of the geothermal loop, causing deviations from the intended design during operation [71]. Furthermore, once the pile is completed and the upper structure starts construction, it will not be possible to repair or replace the pipeline of the energy pile. Any operational discrepancies arising from improper construction can only be addressed by integrating other systems.



Fig. 12. Time period of sustainability aspects of energy saving measures [133].

7.1.3. Operation perspective

Current research lacks a comprehensive sensitivity analysis of energy piles and the application of rational optimization technique for their operation. The coupling of thermal and mechanical effects, as well as the interaction between piles and soil, presents a complex scenario. Studies indicate that the sensitivity of numerical prediction of energy piles is much more complex than previously thought by Caulk et al. [144]. With changes in external forcing such as mechanical loads or temperature variations, the impact of input factors (such as model parameters) on the prediction of different spatial aggregated model outputs will change [145]. However, there is currently little research on global sensitivity analysis of energy piles.

Operational optimization has demonstrated the potential to further improve the thermal performance of energy piles. Strategies such as zoning operation [86] or intermittent operation [85] can both improve system efficiency by mitigating soil thermal imbalance. However, current optimization studies often lack the use of specific optimization methods. Instead, they compare performance by setting different fixed values (e.g., comparing system efficiency under 8 h, 16 h, 24 h operation times). Such an optimization does not consider all solutions and the result may not be optimal.

7.1.4. Assessment perspective

There is a lack of comprehensive evaluation for the long-term operation of energy pile systems. Existing literature mostly focuses on the short-term thermal effects of energy piles, but the long-term operation of the system will inevitably lead to changes in its thermal efficiency [24]. Therefore, monitoring and evaluating the long-term operation of energy pile systems are crucial. Furthermore, a significant limitation to the widespread adoption of energy pile technology is the high initial investment cost. Conversely, its environmentally friendly characteristics offer substantial benefits [24]. Therefore, a comprehensive evaluation considering economic output and environmental benefits is essential.

Current literature on environmental analysis of energy piles predominantly focuses on the greenhouse gas emission reduction effect during the operation of energy piles but lacks a comprehensive evaluation of the environmental impact throughout their entire lifecycle. Energy piles have an impact on the environment during their manufacturing, installation, and operation stages. A comprehensive assessment is needed from the perspective of the entire lifecycle.

7.2. Future research and development of energy piles for carbon reduction

7.2.1. Design perspective

To advance the design of energy piles, it is essential to employ more design optimization techniques for exploring optimal design variables. In particular, Pareto-based methods and decomposition-based methods [146] can be used for solving MOO problems in order to fill the research gap in this direction.

There is a need for the development of specific guidelines and a compilation of typical case references for energy pile design applicable to different regions. In Ref. [24,134], it is evident that the energy, environmental, and economic benefits of energy pile systems are significantly influenced by the climate zone. For instance, applying energy piles in Harbin, located in severe cold zone, will contribute to 45 % energy saving, 40 % utility cost-saving, and 36 % CO₂ emission reduction. However, in Kunming, which is located in mild zone, the energy saving, utility cost-saving and CO₂ emission reduction will be 19 %, 21 %, 26 % respectively when the energy piles are used [24].

The construction and continuous improvement of an integrated design platform are essential. This platform should facilitate the incorporation of optimization algorithms, allowing for real-time adjustment of parameters and simulation of energy pile performance under various operating conditions.

7.2.2. Construction perspective

Lean construction principles have shown promise in maximizing value and minimizing waste, as seen in Ref. [147,148], where lean construction was implemented in the construction process of a steel structure building, resulting in a significant reduction of 43 tons of carbon lean construction [148]. It would be beneficial to explore the application of "lean principles" specifically in the construction of energy piles to assess the potential carbon reduction effects.

Encouraging research on prefabricated energy piles is essential. In the materialization stage, the production of building materials typically generates significant carbon emissions [149]. The implementation of prefabrication has the potential to reduce construction waste, increase material recycling rates, and consequently lower carbon emissions [150–152].

Further studies are necessary to investigate construction methods for different types of energy piles to prevent a loss of system efficiency due to improper construction. The development of a standardized construction manual based on successful existing experiences would contribute to the effective construction of energy piles.

7.2.3. Operation perspective

In terms of single pile operation, it is crucial to focus on key scientific issues such as "heat transfer mechanism of energy piles under air–soil coupling boundary" and "heat transfer basic theory of energy pilesbuilding coupling". A deeper understanding of these theories can provide valuable directions for optimization, and serve as a theoretical foundation for the development of software related to energy pile design.

From the perspective of pile group system operation, existing research on the operational optimization of energy pile groups has been

limited to a simple approach based on a set of comparative values. To achieve an optimal solution, more sophisticated optimization techniques need to be applied, addressing the complexities inherent in the operation of pile groups.

From the perspective of overall system operation, considering that the heat load provided by energy piles is constrained by the size of the pile group, further investigation into hybrid systems combining energy piles with other HVAC technologies is essential to meet the energy demand of buildings. Additionally, research on control optimization and the application effects (e.g., energy efficiency, economic impact, and environmental impact) of hybrid systems is required.

7.2.4. Assessment perspective

The establishment of a standardized evaluation system that incorporates considerations for energy savings, environmental impact, and economic benefits is crucial for a comprehensive assessment of the advantages associated with energy piles. This standardized approach would serve to garner increased attention in this field. Moreover, there is a pronounced need for the routine collection and analysis of data pertaining to energy pile applications and case studies. The scarcity of literature detailing long-term operational monitoring results for energy piles and corresponding analyses of the economic and environmental benefits underscores the necessity for more research in this area. The integration of such data is paramount to comprehending the trends and limitations of energy pile technology applications.

8. Conclusions

This paper provides a comprehensive critical review on energy pile design, construction, operation, and assessment, particularly focusing on carbon reduction from a life-cycle perspective. The integration of life cycle assessment and existing case studies has demonstrated the advantages of optimizing energy piles for carbon reduction at different stages of their life cycle. Several key conclusions and suggestions of this study include:

- The energy efficiency of energy piles presents a substantial advantage in carbon reduction, achieving a noteworthy 75 % reduction in electricity consumption for the same energy output. This efficiency holds significant potential for making a substantial impact on carbon reduction efforts. However, it is crucial to calculate the actual carbon reduction considering the intricacies of the local energy structure.
- To maximize the carbon reduction potential of energy piles, it is recommended to employ advanced optimization techniques during both the design and operation processes. This involves moving beyond simplistic comparisons based on predetermined values for design and operational factors, enabling a more tailored and effective approach to achieve optimal efficiency.
- The construction stage and the selection of building materials for energy piles are frequently disregarded in terms of carbon emissions. To enhance their carbon reduction potential, integrating "lean principles" into the construction phase can optimize processes and minimize embodied carbon emissions. Additionally, exploring the use of low-carbon concrete for energy pile foundations holds promise for achieving a significant reduction in embodied carbon emissions.
- Despite a scarcity of comprehensive evaluations of energy piles, existing studies consistently confirm their substantial environmental advantages over conventional HVAC systems. To encourage wider application and awareness of energy piles, it is crucial to conduct more thorough evaluations that consider various factors influencing their environmental impacts.
- Considering the current global emphasis on carbon reduction and the broad applicability and advantages of energy piles compared to other energy technologies, their implementation for reducing the carbon emissions of HVAC systems holds tremendous promise. Energy piles present significant advantages in terms of carbon reduction benefits,

and by addressing the recommendations outlined above and capitalizing on the global momentum for carbon reduction, the prospects for their widespread adoption in HVAC systems are exceptionally promising.

CRediT authorship contribution statement

Linfeng Zhang: Conceptualization, Formal analysis, Funding acquisition, Investigation, Methodology, Writing – original draft. Haozhe Han: Conceptualization, Formal analysis, Investigation, Methodology, Writing – original draft. Wenxin Li: Conceptualization, Formal analysis, Investigation, Methodology, Writing – review & editing. Kewei Guo: Conceptualization, Formal analysis, Investigation. Minglu Yuan: Conceptualization, Formal analysis, Investigation. Zhengxuan Liu: Conceptualization, Formal analysis, Funding acquisition, Investigation, Methodology, Supervision, Writing – original draft, Writing – review & editing.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

No data was used for the research described in the article.

Acknowledgements

This work was supported by the National Natural Science Foundation of China (Grant number: No. 52008088 & No. 52208106), The Carbon Emission Peak and Carbon Neutrality Innovative Science Foundation of Jiangsu Province "The key research and demonstration projects of future low-carbon emission buildings" (No. BE2022606).

References

- [1] Y. Zhou, Z. Liu, A cross-scale 'material-component-system' framework for transition towards zero-carbon buildings and districts with low, medium and high-temperature phase change materials, Sustain. Cities Soc. 89 (2023) 104378, https://doi.org/10.1016/j.scs.2022.104378.
- [2] Y. Zhou, S. Zheng, Z. Liu, T. Wen, Z. Ding, J. Yan, et al., Passive and active phase change materials integrated building energy systems with advanced machinelearning based climate-adaptive designs, intelligent operations, uncertaintybased analysis and optimisations: A state-of-the-art review, Renew. Sustain. Energy Rev. 130 (2020) 109889, https://doi.org/10.1016/j.rser.2020.109889.
- [3] United Nations Environment Programme. Emissions Gap Emissions Gap Report 2020. 2020.
- [4] UNEP-hosted Global Alliance for Buildings and Construction. 2022 Global Status Report for Buildings and Construction: Towards a Zero-emission, Efficient and Resilient Buildings and Construction Sector. Airobi: 2022.
- [5] Cai W, Wu Y, Ni J, Yu Y, Wu J. 2022 Research Report of China Building Energy Consumption and Carbon Emissions. Chongqing: 2022.
- [6] Z. Liu, Y. Sun, C. Xing, J. Liu, Y. He, Y. Zhou, et al., Artificial intelligence powered large-scale renewable integrations in multi-energy systems for carbon neutrality transition: Challenges and future perspectives, Energy AI 10 (2022) 100195, https://doi.org/10.1016/j.egyai.2022.100195.
- [7] K. You, Y. Yu, W. Cai, Z. Liu, The change in temporal trend and spatial distribution of CO2 emissions of China's public and commercial buildings, Build. Environ. 229 (2023) 109956, https://doi.org/10.1016/j.buildenv.2022.109956.
- [8] Z. Liu, Z(Jerry). Yu, T. Yang, M. El Mankibi, L. Roccamena, Y. Sun, et al., Experimental and numerical study of a vertical earth-to-air heat exchanger system integrated with annular phase change material, Energy Convers Manag 186 (2019) 433–449, https://doi.org/10.1016/j.enconman.2019.02.069.
- [9] L. Zhang, J. Chen, J. Wang, G. Huang, Estimation of soil and grout thermal properties for ground-coupled heat pump systems: Development and application, Appl. Therm. Eng. 143 (2018) 112–122, https://doi.org/10.1016/j. applthermaleng.2018.07.089.
- [10] Z. Liu, M. Xie, Y. Zhou, Y. He, L. Zhang, G. Zhang, et al., A state-of-the-art review on shallow geothermal ventilation systems with thermal performance enhancement[sbnd] system classifications, advanced technologies and applications, Energy Built. Environ. 4 (2023) 148–168, https://doi.org/10.1016/ j.enbenv.2021.10.003.

- [11] O.A. Mustafa, Ground-source heat pumps systems and applications, Renew. Sustain. Energy Rev. 12 (2008) 344–371, https://doi.org/10.1016/j. rser.2006.10.003.
- [12] L. Bandeira, G. Narsilio, N. Makasis, R. Choudhary, Y. Carden, Experimental and numerical data of thermal response tests executed in groups of energy piles connected in series 2023;48. https://doi.org/10.1016/j.dib.2023.109256.
- [13] S. Cui, C. Zhou, J. Liu, D. Bashir, Computers and Geotechnics Stress effects on thermal conductivity of soils and heat transfer efficiency of energy piles in the saturated and unsaturated soils, Comput. Geotech. 160 (2023) 105549, https:// doi.org/10.1016/j.compgeo.2023.105549.
- [14] Y. Cui, J. Zhu, F. Meng, Techno-economic evaluation of multiple energy piles for a ground-coupled heat pump system, Energy Convers Manag 178 (2018) 200–216, https://doi.org/10.1016/j.enconman.2018.10.042.
- [15] B. Morrone, G. Coppola, V. Raucci, Energy and economic savings using geothermal heat pumps in different climates, Energy Convers Manag 88 (2014) 189–198, https://doi.org/10.1016/j.enconman.2014.08.007.
- [16] G. Anis Akrouch, M. Sánchez, J.L. Briaud, Thermal performance and economic study of an energy piles system under cooling dominated conditions, Renew. Energy 147 (2020) 2736–2747, https://doi.org/10.1016/j.renene.2018.11.101.
- [17] A.K. Sani, R.M. Singh, Response of unsaturated soils to heating of geothermal energy pile, Renew. Energy 147 (2020) 2618–2632, https://doi.org/10.1016/j. renene.2018.11.032.
- [18] S. Park, S. Lee, C. Sung, H. Choi, Applicability evaluation of cast-in-place energy piles based on two-year heating and cooling operation, Renew. Sustain. Energy Rev. 143 (2021), https://doi.org/10.1016/j.rser.2021.110906.
- [19] Z. Mohamad, F. Fardoun, F. Meftah, A review on energy piles design, evaluation, and optimization, J. Clean. Prod. 292 (2021), https://doi.org/10.1016/j. iclepro.2021.125802.
- [20] A.K. Sani, R.M. Singh, T. Amis, I. Cavarretta, A review on the performance of geothermal energy pile foundation, its design process and applications, Renew. Sustain. Energy Rev. 106 (2019) 54–78, https://doi.org/10.1016/j. rser.2019.02.008.
- [21] J. Fadejev, R. Simson, J. Kurnitski, F. Haghighat, A review on energy piles design, sizing and modelling, Energy 122 (2017) 390–407, https://doi.org/10.1016/j. energy.2017.01.097.
- [22] M. De Moel, P.M. Bach, A. Bouazza, R.M. Singh, J.O. Sun, Technological advances and applications of geothermal energy pile foundations and their feasibility in Australia, Renew. Sustain. Energy Rev. 14 (2010), https://doi.org/10.1016/j. rser.2010.07.027.
- [23] R.P. Cunha, P.J. Bourne-Webb, A critical review on the current knowledge of geothermal energy piles to sustainably climatize buildings, Renew. Sustain. Energy Rev. 158 (2022), https://doi.org/10.1016/j.rser.2022.112072.
- [24] C. Han, C. Zhu, Y. Shen, X.B. Yu, Energy, environmental and economic performance evaluation of energy pile system under different climate conditions, Energy Convers Manag 252 (2022), https://doi.org/10.1016/j. encomman.2021.115041.
- [25] S. Park, C. Sung, K. Jung, B. Sohn, A. Chauchois, H. Choi, Constructability and heat exchange efficiency of large diameter cast-in-place energy piles with various configurations of heat exchange pipe, Appl. Therm. Eng. 90 (2015) 1061–1071, https://doi.org/10.1016/j.applthermaleng.2015.05.044.
- [26] S. Park, D. Lee, H.J. Choi, K. Jung, H. Choi, Relative constructability and thermal performance of cast-in-place concrete energy pile: Coil-type GHEX (ground heat exchanger), Energy 81 (2015) 56–66, https://doi.org/10.1016/j. energy.2014 08.012.
- [27] S. Park, D. Lee, S. Lee, A. Chauchois, H. Choi, Experimental and numerical analysis on thermal performance of large-diameter cast-in-place energy pile constructed in soft ground, Energy 118 (2017) 297–311, https://doi.org/ 10.1016/j.energy.2016.12.045.
- [28] M.A. Jalaluddin, K. Tsubaki, S. Inoue, K. Yoshida, Experimental study of several types of ground heat exchanger using a steel pile foundation. Renew, Energy 36 (2011) 764–771, https://doi.org/10.1016/j.renene.2010.08.011.
- [29] A. Miyara, K. Tsubaki, K. Yoshida, Thermal performances of three types of ground heat exchangers in short-time period of operation, Refrig Air Cond (2010) 1–8.
- [30] K. Morino, T. Oka, Study on heat exchanged in soil by circulating water in a steel pile, Energ. Build. 21 (1994) 65–78, https://doi.org/10.1016/0378-7788(94) 90017-5.
- [31] C. de Freitas, M. Murari, C. de Hollanda Cavalcanti Tsuha, F. Loveridge, Investigation on the thermal response of steel pipe energy piles with different backfill materials, Renew. Energy 199 (2022) 44–61, https://doi.org/10.1016/j. renene.2022.08.105.
- [32] Y. Hamada, H. Saitoh, M. Nakamura, H. Kubota, K. Ochifuji, Field performance of an energy pile system for space heating, Energ. Build. 39 (2007) 517–524, https://doi.org/10.1016/j.enbuild.2006.09.006.
- [33] H. Park, S.R. Lee, S. Yoon, J.C. Choi, Evaluation of thermal response and performance of PHC energy pile: Field experiments and numerical simulation, Appl. Energy 103 (2013) 12–24, https://doi.org/10.1016/j. apenergy.2012.10.012.
- [34] J. Luo, H. Zhao, S. Gui, W. Xiang, J. Rohn, P. Blum, Thermo-economic analysis of four different types of ground heat exchangers in energy piles, Appl. Therm. Eng. 108 (2016) 11–19, https://doi.org/10.1016/j.applthermaleng.2016.07.085.
- [35] J. Gao, X. Zhang, J. Liu, K.S. Li, J. Yang, Thermal performance and ground temperature of vertical pile-foundation heat exchangers: A case study, Appl. Therm. Eng. 28 (2008) 2295–2304, https://doi.org/10.1016/j. applthermaleng.2008.01.013.

- [36] Ministry of Housing and Urban-Rural Development of the People's Republic of China. Technical standards for utilization of geothermal energy through piles JGJT438-2018. 2018.
- [37] A.A. Mehrizi, S. Porkhial, B. Bezyan, H. Lotfizadeh, Energy pile foundation simulation for different configurations of ground source heat exchanger, Int. Commun. Heat Mass Transf. 70 (2016) 105–114, https://doi.org/10.1016/j. icheatmasstransfer.2015.12.001.
- [38] J. Gao, X. Zhang, J. Liu, K. Li, J. Yang, Numerical and experimental assessment of thermal performance of vertical energy piles: An application, Appl. Energy 85 (2008) 901–910, https://doi.org/10.1016/j.apenergy.2008.02.010.
- [39] A. Zarrella, M. De Carli, A. Galgaro, Thermal performance of two types of energy foundation pile: Helical pipe and triple U-tube, Appl. Therm. Eng. 61 (2013) 301–310, https://doi.org/10.1016/j.applthermaleng.2013.08.011.
- [40] O. Khandouzi, M. Pourfallah, E. Yoosefirad, B. Shaker, M. Gholinia, S. Mouloodi, Evaluating and optimizing the geometry of thermal foundation pipes for the utilization of the geothermal energy: Numerical simulation, J. Energy Storage 37 (2021), https://doi.org/10.1016/j.est.2021.102464.
- [41] S. Yoon, S.R. Lee, J. Xue, K. Zosseder, G.H. Go, H. Park, Evaluation of the thermal efficiency and a cost analysis of different types of ground heat exchangers in energy piles, Energy Convers. Manag. 105 (2015) 393–402, https://doi.org/ 10.1016/j.enconman.2015.08.002.
- [42] M. Faizal, A. Bouazza, R.M. Singh, Heat transfer enhancement of geothermal energy piles, Renew. Sustain. Energy Rev. 57 (2016) 16–33, https://doi.org/ 10.1016/j.rser.2015.12.065.
- [43] S. Lee, S. Park, M. Kang, K. Oh, H. Choi, Effect of tube-in-tube configuration on thermal performance of coaxial-type ground heat exchanger, Renew. Energy 197 (2022) 518–527, https://doi.org/10.1016/j.renene.2022.07.088.
- [44] W. Liu, M. Xu, Research Progress of Pile Heat Exchangers in Ground Source Heat Pump System. Procedia Eng., vol. 205, Elsevier Ltd; 2017, p. 3775–81. https ://doi.org/10.1016/j.proeng.2017.10.141.
- [45] S. Park, S. Lee, D. Lee, D. Ahn, H. Choi, Effect of thermal interference on energy piles considering various configurations of heat exchangers, Energ. Buildings 199 (2019) 381–401, https://doi.org/10.1016/j.enbuild.2019.07.008.
- [46] X. Ding, C. Peng, C. Wang, G. Kong, Heat transfer performance of energy piles in seasonally frozen soil areas, Renew. Energy 190 (2022) 903–918, https://doi. org/10.1016/j.renene.2022.04.004.
- [47] T. You, H. Yang, Feasibility of ground source heat pump using spiral coil energy piles with seepage for hotels in cold regions, Energy Convers Manag 205 (2020), https://doi.org/10.1016/j.enconman.2020.112466.
- [48] H. Song, H. Pei, D. Xu, C. Cui, Performance study of energy piles in different climatic conditions by using multi-sensor technologies, Meas. J. Int. Meas. Confed 162 (2020), https://doi.org/10.1016/j.measurement.2020.107875.
- [49] M. Saaly, P. Maghoul, M. Kavgic, D. Polyzois, Performance analysis of a proposed geothermal pile system for heating and cooling energy demand for a building in cold regions, Sustain. Cities Soc. 45 (2019) 669–682, https://doi.org/10.1016/j. scs.2018.12.014.
- [50] C.A. Kramer, O. Ghasemi-Fare, P. Basu, Laboratory thermal performance tests on a model heat exchanger pile in sand, Geotech. Geol. Eng. 33 (2015) 253–271, https://doi.org/10.1007/s10706-014-9786-z.
- [51] O. Ghasemi-Fare, P. Basu, A practical heat transfer model for geothermal piles, Energ. Buildings 66 (2013) 470–479, https://doi.org/10.1016/j. enbuild.2013.07.048.
- [52] T.M. Bodas Freitas, M. Zito, P.J. Bourne-Webb, D. Sterpi, Thermal performance of seasonally, thermally-activated floating pile foundations in a cohesive medium, Eng. Struct. 243 (2021), https://doi.org/10.1016/j.engstruct.2021.112588.
- [53] Y. Lou, P. fei Fang, X. yu Xie, C.S.A. Chong, F. yuan Li, C. yang Liu, et al., Numerical research on thermal response for geothermal energy pile groups under groundwater flow, Geomech. Energy Environ. 28 (2021). https://doi.org/10.10 16/j.gete.2021.100257.
- [54] D. Wang, L. Lu, W. Zhang, P. Cui, Numerical and analytical analysis of groundwater influence on the pile geothermal heat exchanger with cast-in spiral coils, Appl. Energy 160 (2015) 705–714, https://doi.org/10.1016/j. apenergy.2015.04.037.
- [55] W. Yang, L. Ju, L. Zhang, F. Wang, Experimental investigations of the thermomechanical behaviour of an energy pile with groundwater seepage, Sustain. Cities Soc. 77 (2022) 103588, https://doi.org/10.1016/j.scs.2021.103588.
 [56] E.H.N. Gashti, M. Malaska, K. Kujala, Analysis of thermo-active pile structures
- [56] E.H.N. Gashti, M. Malaska, K. Kujala, Analysis of thermo-active pile structures and their performance under groundwater flow conditions, Energ. Buildings 105 (2015) 1–8, https://doi.org/10.1016/j.enbuild.2015.07.026.
- [57] G.H. Go, S.R. Lee, S. Yoon, H. byul Kang, Design of spiral coil PHC energy pile considering effective borehole thermal resistance and groundwater advection effects, Appl. Energy 125 (2014) 165–178, https://doi.org/10.1016/j. apenergy.2014.03.059.
- [58] T. You, H. Yang, Influences of different factors on the three-dimensional heat transfer of spiral-coil energy pile group with seepage, Int. J. Low-Carbon Technol. 15 (2020) 458–470, https://doi.org/10.1093/IJLCT/CTAA006.
- [59] C.R. de Almeida, C.A. Neto, C. de H.C. Tsuha, M.E.G. Boscov, Numerical study on heat transfer performance of geothermal piles in a Brazilian sandy soil, Soils Rocks 2022;45. https://doi.org/10.28927/SR.2022.076621.
- [60] B.C. Kwag, M. Krarti, Performance of thermoactive foundations for commercial buildings, J. Sol. Energy Eng. Trans. ASME 135 (2013), https://doi.org/10.1115/ 1.4025587.
- [61] F. Cecinato, F.A. Loveridge, Influences on the thermal efficiency of energy piles, Energy 82 (2015) 1021–1033, https://doi.org/10.1016/j.energy.2015.02.001.
- [62] A. Carotenuto, P. Marotta, N. Massarotti, A. Mauro, G. Normino, Energy piles for ground source heat pump applications: Comparison of heat transfer performance

for different design and operating parameters, Appl. Therm. Eng. 124 (2017) 1492–1504, https://doi.org/10.1016/j.applthermaleng.2017.06.038.

- [63] E.H.N. Gashti, V.M. Uotinen, K. Kujala, Numerical modelling of thermal regimes in steel energy pile foundations: A case study, Energ. Buildings 69 (2014) 165–174, https://doi.org/10.1016/j.enbuild.2013.10.028.
- [64] R.Y.W. Liu, E. Sailer, D.M.G. Taborda, D.M. Potts, Evaluating the impact of different pipe arrangements on the thermal performance of thermo-active piles. E3S Web Conf., vol. 205, EDP Sciences; 2020. https://doi.org/10.1051/e3sconf/2 02020505006.
- [65] A.K. Sani, R.M. Singh, C. de H.C. de Tsuha, I. Cavarretta, Pipe-pipe thermal interaction in a geothermal energy pile, Geothermics 81 (2019) 209–223, https:// doi.org/10.1016/j.geothermics.2019.05.004.
- [66] S. Park, S. Lee, D. Lee, S.S. Lee, H. Choi, Influence of coil pitch on thermal performance of coil-type cast-in-place energy piles, Energ. Buildings 129 (2016) 344–356, https://doi.org/10.1016/j.enbuild.2016.08.005.
- [67] A.H. Farajollahi, B. Asgari, M. Rostami, Thermal performance analysis of an energy pile with triple helix ground heat exchanger, Geothermics 104 (2022), https://doi.org/10.1016/j.geothermics.2022.102459.
- [68] W. Lyu, H. Pu, J.N. Chen, Thermal performance of an energy pile group with a deeply penetrating u-shaped heat exchanger, Energies 13 (2020), https://doi. org/10.3390/en13215822.
- [69] T. You, X. Li, S. Cao, H. Yang, Soil thermal imbalance of ground source heat pump systems with spiral-coil energy pile groups under seepage conditions and various influential factors, Energy Convers Manag 178 (2018) 123–136, https://doi.org/ 10.1016/j.encomman.2018.10.027.
- [70] M. Alberdi-Pagola, S.E. Poulsen, R.L. Jensen, S. Madsen, A case study of the sizing and optimisation of an energy pile foundation (Rosborg, Denmark), Renew. Energy 147 (2020) 2724–2735, https://doi.org/10.1016/j.renene.2018.07.100.
- [71] T. Amis, J.S. McCartney, F. Loveridge, G.C. Olgun, M.E. Bruce, K. Murphy, Identifying best practice, installation, laboratory testing and field testing, DFI J. – J. Deep Found Inst. 8 (2014) 74–83, https://doi.org/10.1179/ 1937525514y.000000006.
- [72] F. Loveridge, F. Cecinato, Thermal performance of thermoactive continuous flight auger piles, Environ. Geotech. 3 (2016) 265–279, https://doi.org/10.1680/ jenge.15.00023.
- [73] M.H. Bahmani, O.A. Akbari, M. Zarringhalam, G. Ahmadi Sheikh Shabani, M. Goodarzi, Forced convection in a double tube heat exchanger using nanofluids with constant and variable thermophysical properties, Int. J. Numer. Meth. Heat Fluid Flow 30 (2020) 3247–3265, https://doi.org/10.1108/HFF-01-2019-0017.
- [74] M. Goodarzi, A. Amiri, M.S. Goodarzi, M.R. Safaei, A. Karimipour, E.M. Languri, et al., Investigation of heat transfer and pressure drop of a counter flow corrugated plate heat exchanger using MWCNT based nanofluids, Int. Commun. Heat Mass Transf. 66 (2015) 172–179, https://doi.org/10.1016/j. icheatmasstransfer.2015.05.002.
- [75] M.M. Sarafraz, M.R. Safaei, Z. Tian, M. Goodarzi, E.P.B. Filho, M. Arjomandi, Thermal assessment of nano-particulate graphene-water/ethylene glycol (WEG 60:40) nano-suspension in a compact heat exchanger, Energies 12 (2019), https://doi.org/10.3390/en12101929.
- [76] L. peng Kong, L. Qiao, Y. yang Xiao, Q. wen Li, A study on heat transfer characteristics and pile group influence of enhanced heat transfer energy piles, J. Build. Eng. (2019) 24. https://doi.org/10.1016/j.jobe.2019.100768.
- [77] Y. Shen, B. Qu, K. Chen, C. Gong, C. Han, Environmental and energy benefits assessment of alkali-activated concrete (AAC) energy pile, J. Build. Eng. 61 (2022), https://doi.org/10.1016/j.jobe.2022.105282.
- [78] S. Lee, S. Park, D. Ahn, H. Choi, Thermal performance of novel cast-in-place energy piles equipped with multipurpose steel pipe heat exchangers (SPHXs), Geothermics 102 (2022), https://doi.org/10.1016/j.geothermics.2022.102389.
- [79] K. Ahmed, M. Al-Khawaja, M. Suleiman, Optimization of energy pile conductance using finite element and fractional factorial design of experiment. IOP Conf. Ser. Mater. Sci. Eng., vol. 383, Institute of Physics Publishing; 2018. https://doi.org/1 0.1088/1757-899X/383/1/012034.
- [80] X. Bao, X. Qi, H. Cui, J. Zou, X. Xiao, Comparison study on the performance of a novel and traditional energy piles by laboratory tests. Symmetry (Basel) 2021;13. https://doi.org/10.3390/sym13101958.
- [81] Z. Cao, G. Zhang, Y. Liu, X. Zhao, Thermal performance analysis and assessment of PCM backfilled precast high-strength concrete energy pile under heating and cooling modes of building, Appl. Therm. Eng. 216 (2022), https://doi.org/ 10.1016/j.applthermaleng.2022.119144.
- [82] R. Li, G. Kong, Y. Chen, Q. Yang, Thermomechanical behaviour of an energy pile–raft foundation under intermittent cooling operation. Geomech, Energy Environ. 28 (2021), https://doi.org/10.1016/j.gete.2021.100240.
 [83] Z. Cao, G. Zhang, Y. Liu, X. Zhao, C. Li, Influence of backfilling phase change
- [83] Z. Cao, G. Zhang, Y. Liu, X. Zhao, C. Li, Influence of backfilling phase change material on thermal performance of precast high-strength concrete energy pile, Renew. Energy 184 (2022) 374–390, https://doi.org/10.1016/j. renene.2021.11.100.
- [84] Y. Cui, J. Zhu, CFD assessment of multiple energy piles for ground source heat pump in heating mode, Appl. Therm. Eng. 139 (2018) 99–112, https://doi.org/ 10.1016/j.applthermaleng.2018.04.073.
- [85] M. Faizal, A. Bouazza, R.M. Singh, An experimental investigation of the influence of intermittent and continuous operating modes on the thermal behaviour of a full scale geothermal energy pile, Geomech. Energy Environ. 8 (2016) 8–29, https://doi.org/10.1016/j.gete.2016.08.001.
- [86] T. You, W. Zeng, Zoning operation of energy piles to alleviate the soil thermal imbalance of ground source heat pump systems, Energy Built. Environ. 4 (2023) 57–63, https://doi.org/10.1016/j.enbenv.2021.08.001.

- [87] Z. Ma, L. Xia, X. Gong, G. Kokogiannakis, S. Wang, X. Zhou, Recent advances and development in optimal design and control of ground source heat pump systems, Renew. Sustain. Energy Rev. 131 (2020), https://doi.org/10.1016/j. rser.2020.110001.
- [88] H. Al Moussawi, F. Fardoun, H. Louahlia-Gualous, Review of tri-generation technologies: Design evaluation, optimization, decision-making, and selection approach, Energy Convers Manag 120 (2016) 157–196, https://doi.org/10.1016/ j.enconman.2016.04.085.
- [89] K. Ahmed, M. Al-khawaja, M. Suleiman, Uniform fractional factorial design tables for energy piles with maximum 2017;224:175–86. https://doi.org/10.2495/E SUS170171.
- [90] S. Haridy, K. Alnaqbi, A. Radwan, M.G. Arab, Optimizing the thermal performance of energy piles using response surface methodology. Case Stud, Therm. Eng. 41 (2023), https://doi.org/10.1016/j.csite.2022.102637.
- [91] R. Nazmabadi, B. Asgari, A. Hakkaki-Fard, On the thermal performance enhancement of spiral-coil energy piles with a thermal recovery system, Energ. Buildings 269 (2022), https://doi.org/10.1016/j.enbuild.2022.112240.
- [92] B. Bezyan, S. Porkhial, A.A. Mehrizi, 3-D simulation of heat transfer rate in geothermal pile-foundation heat exchangers with spiral pipe configuration, Appl. Therm. Eng. 87 (2015) 655–668, https://doi.org/10.1016/j. applthermaleng.2015.05.051.
- [93] Q. Li, L. Chen, H. Ma, C.H. Huang, Enhanced heat transfer characteristics of graphite concrete and its application in energy piles, Adv. Mater. Sci. Eng. 2018 (2018), https://doi.org/10.1155/2018/8142392.
- [94] O. Elkezza, M. Mohamed, A. Khan, Performance of thermally enhanced geoenergy piles and walls, Geothermics 98 (2022), https://doi.org/10.1016/j. geothermics.2021.102274.
- [95] G. Huang, Z. Liao, S. Li, C. Zhuang, H. Zhang, L. Cheng, et al., A novel independent heat extraction-release double helix energy pile: Numerical and experimental investigations of heat extraction effect, Energy Convers Manag 254 (2022), https://doi.org/10.1016/j.enconman.2022.115249.
- [96] Z. Chen, J. Yao, P. Pan, H. Xiao, Q. Ma, Research on the heat exchange characteristics of the deeply buried pipe type of energy pile, Case Stud. Therm. Eng. 27 (2021), https://doi.org/10.1016/j.csite.2021.101268.
- [97] W. Lyu, H. Pu, H. Xiao, D. Hu, Q. Ma, Thermal performance of energy pile with deeply penetrating 1-U-shape heat exchanger, Geothermics 91 (2021), https:// doi.org/10.1016/j.geothermics.2020.102023.
- [98] H. Javadi, S.S.M. Ajarostaghi, S.S. Mousavi, M. Pourfallah, Thermal analysis of a triple helix ground heat exchanger using numerical simulation and multiple linear regression, Geothermics 81 (2019) 53–73, https://doi.org/10.1016/j. geothermics.2019.04.005.
- [99] J. Huang, J.S. McCartney, H. Perko, D. Johnson, C. Zheng, Q. Yang, A novel energy pile: The thermo-syphon helical pile, Appl. Therm. Eng. 159 (2019). htt ps://doi.org/10.1016/j.applthermaleng.2019.113882.
- [100] G. Huang, X. Yang, Y. Liu, C. Zhuang, H. Zhang, J. Lu, A novel truncated cone helix energy pile: Modelling and investigations of thermal performance, Energ. Buildings 158 (2018) 1241–1256, https://doi.org/10.1016/j. enbuild.2017.11.020.
- [101] Q. Li, Y. Yin, L. Kong, L. Qiao, Enhancing heat transfer in the heat exchange medium of energy piles, J. Build. Eng 40 (2021), https://doi.org/10.1016/j. jobe.2021.102375.
- [102] K. Meng, C. Cui, H. Li, H. Liu, Ontology-based approach supporting multiobjective holistic decision making for energy pile system, Buildings 12 (2022), https://doi.org/10.3390/buildings12020236.
- [103] F. Cecinato, D. Salciarini, Energy performance assessment of thermo-active micropiles via numerical modeling and statistical analysis, Geomech. Energy Environ. 29 (2022), https://doi.org/10.1016/j.gete.2021.100268.
- [104] S.T. Ng, J.M.W. Wong, S. Skitmore, A. Veronika, Carbon dioxide reduction in the building life cycle: A critical review, Proc. Inst. Civ. Eng. Eng. Sustain. 165 (2012) 281–292, https://doi.org/10.1680/ensu.11.00005.
- [105] M. Roberts, S. Allen, D. Coley, Life cycle assessment in the building design process – A systematic literature review, Build. Environ. 185 (2020), https://doi.org/ 10.1016/j.buildenv.2020.107274.
- [106] H. Brandl, Thermo-active ground-source structures for heating and cooling. Procedia Eng., vol. 57, Elsevier Ltd; 2013, p. 9–18. https://doi.org/10.1016/j. proeng.2013.04.005.
- [107] C.G. Olgun, J.S. McCartney, Outcomes from international workshop on thermoactive geotechnical systems for near-surface geothermal energy: from research to practice, DFI J – J. Deep Found Inst. 8 (2014) 59–73, https://doi.org/ 10.1179/1937525514y.0000000005.
- [108] H. Brandl, Energy foundations and other thermo-active ground structures, Geotechnique 56 (2006) 81–122, https://doi.org/10.1680/geot.2006.56.2.81.
- [109] N. Batini, A.F. Rotta Loria, P. Conti, D. Testi, W. Grassi, L. Laloui, Energy and geotechnical behaviour of energy piles for different design solutions, Appl. Therm. Eng. 86 (2015) 199–213, https://doi.org/10.1016/j. applthermaleng.2015.04.050.
- [110] F. Loveridge, W. Powrie, 2D thermal resistance of pile heat exchangers, Geothermics 50 (2014) 122–135, https://doi.org/10.1016/j. geothermics.2013.09.015.
- [111] F. Loveridge, The Thermal Performance of Foundation Piles used as Heat Exchangers in Ground Energy Systems. 2012.
- [112] A.K. Tiwari, A. Kumar, P. Basu, The influence of thermal interaction on energy harvesting efficiency of geothermal piles in a group, Appl. Therm. Eng. 200 (2022), https://doi.org/10.1016/j.applthermaleng.2021.117673.

- [113] X. Bao, X. Qi, H. Cui, W. Tang, X. Chen, Experimental study on thermal response of a PCM energy pile in unsaturated clay, Renew. Energy 185 (2022) 790–803, https://doi.org/10.1016/j.renene.2021.12.062.
- [114] H. Cui, J. Zou, Z. Gong, D. Zheng, X. Bao, X. Chen, Study on the thermal and mechanical properties of steel fibre reinforced PCM-HSB concrete for high performance in energy piles, Constr. Build. Mater. 350 (2022), https://doi.org/ 10.1016/j.conbuildmat.2022.128822.
- [115] M.M. Mousa, A.M. Bayomy, M.Z. Saghir, Long-term performance investigation of a GSHP with actual size energy pile with PCM, Appl. Therm. Eng. 210 (2022), https://doi.org/10.1016/j.applthermaleng.2022.118381.
- [116] M.U. Hossain, C.S. Poon, I.M.C. Lo, J.C.P. Cheng, Comparative LCA on using waste materials in the cement industry: A Hong Kong case study, Resour. Conserv. Recycl. 120 (2017) 199–208, https://doi.org/10.1016/j.resconrec.2016.12.012.
- [117] D.J.M. Flower, J.G. Sanjayan, Green house gas emissions due to concrete manufacture, Int. J. Life Cycle Assess. 12 (2007) 282–288, https://doi.org/ 10.1007/s11367-007-0327-3.
- [118] Zhao H, Sun W, Wu X, Gao B. Sustainable self-compacting concrete containing high-amount industrial by-product fly ash as supplementary cementitious materials n.d. https://doi.org/10.1007/s11356-021-15883-2/Published.
- [119] B.C. McLellan, R.P. Williams, J. Lay, A. Van Riessen, G.D. Corder, Costs and carbon emissions for geopolymer pastes in comparison to ordinary portland cement, J. Clean. Prod. 19 (2011) 1080–1090, https://doi.org/10.1016/j. jclepro.2011.02.010.
- [120] R. Abbas, M.A. Khereby, H.Y. Ghorab, N. Elkhoshkhany, Preparation of geopolymer concrete using Egyptian kaolin clay and the study of its environmental effects and economic cost, Clean Techn. Environ. Policy 22 (2020) 669–687, https://doi.org/10.1007/s10098-020-01811-4.
- [121] S. Nie, J. Zhou, F. Yang, M. Lan, J. Li, Z. Zhang, et al., Analysis of theoretical carbon dioxide emissions from cement production: Methodology and application, J. Clean. Prod. 334 (2022), https://doi.org/10.1016/j.jclepro.2021.130270.
- [122] J.U. Lee, T. Kim, S.B. Leigh, Thermal performance analysis of a ground-coupled heat pump integrated with building foundation in summer, Energ. Buildings 59 (2013) 37–43, https://doi.org/10.1016/j.enbuild.2012.12.016.
- [123] Y. Cui, J. Zhu, Year-round performance assessment of a ground source heat pump with multiple energy piles, Energ. Buildings 158 (2018) 509–524, https://doi. org/10.1016/j.enbuild.2017.10.033.
- [124] C.E. Moon, J.M. Choi, Heating performance characteristics of the ground source heat pump system with energy-piles and energy-slabs, Energy 81 (2015) 27–32, https://doi.org/10.1016/j.energy.2014.10.063.
- [125] L. Aresti, P. Christodoulides, G.P. Panayiotou, G. Florides, Residential buildings' foundations as a ground heat exchanger and comparison among different types in a moderate climate country, Energies 13 (2020), https://doi.org/10.3390/ en13236287.
- [126] C.G. Olgun, T.Y. Ozudogru, S.L. Abdelaziz, A. Senol, Long-term performance of heat exchanger piles, Acta Geotech. 10 (2015) 553–569, https://doi.org/ 10.1007/s11440-014-0334-z.
- [127] Y. Guo, G. Zhang, S. Liu, Investigation on the thermal response of full-scale PHC energy pile and ground temperature in multi-layer strata, Appl. Therm. Eng. 143 (2018) 836–848, https://doi.org/10.1016/j.applthermaleng.2018.08.005.
- [128] L. wei Ren, J. Xu, G. qiang Kong, H. long Liu, Field tests on thermal response characteristics of micro-steel-pipe pile under multiple temperature cycles, Renew. Energy 147 (2020) 1098–1106, https://doi.org/10.1016/j.renene.2019.09.084.
- [129] A. Bowers, A.M. Asce, C. Guney Olgun, M. Asce, Optimization of Energy Pile Grids during Unbalanced Heating and Cooling Operations. n.d.
- [130] S. You, X. Cheng, H. Guo, Z. Yao, In-situ experimental study of heat exchange capacity of CFG pile geothermal exchangers, Energ. Buildings 79 (2014) 23–31, https://doi.org/10.1016/j.enbuild.2014.04.021.
- [131] W. Yang, P. Lu, Y. Chen, Laboratory investigations of the thermal performance of an energy pile with spiral coil ground heat exchanger, Energ. Buildings 128 (2016) 491–502, https://doi.org/10.1016/j.enbuild.2016.07.012.
- [132] H. Ayaz, M. Faizal, A. Bouazza, Thermo-energy performance of neighbouring energy piles. Soils and Rocks 2022;45. https://doi.org/10.28927/SR.2022.07652
- [133] B. Huang, V. Mauerhofer, Life cycle sustainability assessment of ground source heat pump in Shanghai, China. J Clean Prod 119 (2016) 207–214, https://doi. org/10.1016/j.jclepro.2015.08.048.
- [134] M. Sutman, G. Speranza, A. Ferrari, P. Larrey-Lassalle, L. Laloui, Long-term performance and life cycle assessment of energy piles in three different climatic

conditions, Renew. Energy 146 (2020) 1177-1191, https://doi.org/10.1016/j. renene.2019.07.035.

- [135] M. Smith, A. Bevacqua, S. Tembe, P. Lal, Life cycle analysis (LCA) of residential ground source heat pump systems: A comparative analysis of energy efficiency in New Jersey, Sustain Energy Technol Assessments 47 (2021), https://doi.org/ 10.1016/j.seta.2021.101364.
- [136] C.J. Koroneos, E.A. Nanaki, Environmental impact assessment of a ground source heat pump system in Greece, Geothermics 65 (2017) 1–9, https://doi.org/ 10.1016/j.geothermics.2016.08.005.
- [137] H. Song, H. Pei, C. Zhou, D. Zou, C. Cui, Calculation of the representative temperature change for the thermomechanical design of energy piles. Geomech, Energy Environ. 29 (2022), https://doi.org/10.1016/j.gete.2021.100264.
- [138] S. Abdelaziz, T.Y. Ozudogru, Non-uniform thermal strains and stresses in energy piles, Environ. Geotech. 3 (2016) 237–252, https://doi.org/10.1680/ jence.15.00032.
- [139] Y. Cui, J. Zhu, S. Twaha, S. Riffat, A comprehensive review on 2D and 3D models of vertical ground heat exchangers, Renew. Sustain. Energy Rev. 94 (2018) 84–114, https://doi.org/10.1016/j.rser.2018.05.063.
- [140] D.Y. Cherati, O. Ghasemi-Fare, Practical approaches for implementation of energy piles in Iran based on the lessons learned from the developed countries experiences, Renew. Sustain. Energy Rev. 140 (2021), https://doi.org/10.1016/j. rser.2021.110748.
- [141] S. You, C.H. Zhang, P. Jiang, Heat transfer performance and structure response of energy piles. Therm. Sci. 23 (2019) 1647–1654. https://doi.org/10.2298/TSCI1 80602235Y.
- [142] P.J. Bourne-Webb, T.M. Bodas Freitas, Thermally-activated piles and pile groups under monotonic and cyclic thermal loading–A review, Renew. Energy 147 (2020) 2572–2581, https://doi.org/10.1016/j.renene.2018.11.025.
- [143] W. Choi, H. Kikumoto, R. Ooka, New perspectives in thermal performance test: Cost-effective apparatus and extended data analysis, Energ. Buildings 180 (2018) 109–121, https://doi.org/10.1016/j.enbuild.2018.08.008.
- [144] R. Caulk, E. Ghazanfari, J.S. McCartney, Parameterization of a calibrated geothermal energy pile model, Geomech. Energy Environ. 5 (2016) 1–15, https:// doi.org/10.1016/j.gete.2015.11.001.
- [145] D. Shao, G. Jiang, C. Zong, Y. Xing, Z. Zheng, S. Lv, Global sensitivity analysis of behavior of energy pile under thermo-mechanical loads, Soils Found. 61 (2021) 283–302, https://doi.org/10.1016/j.sandf.2020.09.010.
- [146] I. Giagkiozis, P.J. Fleming, Methods for multi-objective optimization: An analysis, Inf. Sci. (Ny) 293 (2015) 338–350, https://doi.org/10.1016/j.ins.2014.08.071.
- [147] S. Wandahl, C.T. Pérez, S. Salling, H.H. Neve, J. Lerche, S. Petersen, The impact of construction labour productivity on the renovation wave, Constr. Econ. Build. 21 (2021) 11–32, https://doi.org/10.5130/AJCEB.V2113.7688.
- [148] P. Fu, J. Sun, C. Pasquire, Carbon emission assessment for steel structure based on lean construction process, J. Intell. Robot. Syst. Theory Appl. 79 (2015) 401–416, https://doi.org/10.1007/s10846-014-0106-x.
- [149] J.L. Hao, B. Cheng, W. Lu, J. Xu, J. Wang, W. Bu, et al., Carbon emission reduction in prefabrication construction during materialization stage: A BIM-based lifecycle assessment approach, Sci. Total Environ. 723 (2020) 137870, https://doi. org/10.1016/j.scitotenv.2020.137870.
- [150] Y. Teng, W. Pan, Systematic embodied carbon assessment and reduction of prefabricated high-rise public residential buildings in Hong Kong, J. Clean. Prod. 238 (2019), https://doi.org/10.1016/j.jclepro.2019.117791.
- [151] V.W.Y. Tam, J.J.L. Hao, Prefabrication as a mean of minimizing construction waste on site, Int. J. Constr. Manag. 14 (2014) 113–121, https://doi.org/ 10.1080/15623599.2014.899129.
- [152] H. Sadeghi, R.M. Singh, Driven precast concrete geothermal energy piles: Current state of knowledge, Build. Environ. 228 (2023), https://doi.org/10.1016/j. buildenv.2022.109790.
- [153] Z. Liu, X. Zhang, Y. Sun, Advanced controls on energy reliability, flexibility and occupant-centric control for smart and energy-efficient buildings, Energ. Buildings 297 (2023), https://doi.org/10.1016/j.enbuild.2023.113436.
- [154] Z. Liu, C. Yu, Q. Qian, R. Huang, K. You, H. Visscher, G. Zhang, Incentive initiatives on energy-efficient renovation of existing buildings towards carbon-neutral blueprints in China: Advancements, challenges and prospects, Eneg. Buildings 296 (2023), https://doi.org/10.1016/j.enbuild.2023.113343.
- [155] Z. Liu, Y. Zhou, J. Yan, M. Tostado-Véliz, Frontier ocean thermal/power and solar PV systems for transformation towards net-zero communities, Energy 284 (2023), https://doi.org/10.1016/j.energy.2023.128362.